# **ECONSTOR** Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Falagiarda, Matteo; McQuade, Peter; Tirpák, Marcel

Working Paper Spillovers from the ECB's non-standard monetary policies on non-euro area EU countries: evidence from an eventstudy analysis

ECB Working Paper, No. 1869

Provided in Cooperation with:

European Central Bank (ECB)

*Suggested Citation:* Falagiarda, Matteo; McQuade, Peter; Tirpák, Marcel (2015) : Spillovers from the ECB's non-standard monetary policies on non-euro area EU countries: evidence from an event-study analysis, ECB Working Paper, No. 1869, ISBN 978-92-899-1682-0, European Central Bank (ECB), Frankfurt a. M., https://doi.org/10.2866/676909

This Version is available at: https://hdl.handle.net/10419/154302

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



## WWW.ECONSTOR.EU

EUROPEAN CENTRAL BANK

## **Working Paper Series**

Matteo Falagiarda, Peter McQuade and Marcel Tirpák Spillovers from the ECB's nonstandard monetary policies on non-euro area EU countries: evidence from an event-study analysis



**Note:** This Working Paper should not be reported as representing the views of the European Central Bank (ECB). The views expressed are those of the authors and do not necessarily reflect those of the ECB

### ABSTRACT

Using event-study techniques we investigate the presence and the magnitude of spillovers from the ECB's non-standard monetary policies on financial assets in selected non-euro area EU countries from Central and Eastern Europe (the Czech Republic, Hungary, Poland and Romania). Generally, we find strong evidence of spillover effects from the ECB's announcements on bond yields. We also find that the SMP announcements resulted in significant spillovers, while those from the OMT and the PSPP announcements were rather limited. Turning to the transmission channels, we argue that spillovers from the SMP announcements went through the portfolio rebalancing and the signalling channels. The transmission of the OMT operated via the confidence channel and for the PSPP we find evidence that both the confidence and the signalling channels were at play.

**Keywords:** monetary policy spillovers, ECB, Central and Eastern Europe, event-study **JEL Classification:** C32, E52, E58

## Non-technical summary

In this paper we examine the international transmission of ECB monetary policy. More specifically, we investigate whether the ECB's non-standard monetary policy measures spilled over to non-euro area EU countries in Central and Eastern Europe (CEE).

The existence of close economic and financial linkages between CEE countries and the euro area has been extensively studied and is well documented in the literature. In general, CEE countries are small and highly open economies for which euro area countries are the dominant trading partners and providers of capital. In addition, banks in CEE countries are to a large extent controlled by banking groups domiciled in the euro area, which has resulted in substantial cross-border banking flows. The very high degree of economic and financial integration between CEE countries and the euro area provides us with an interesting test case for studying potential monetary policy spillovers.

In this paper we investigate whether such monetary spillovers from the euro area to CEE countries following an inflation targeting monetary policy strategy (namely the Czech Republic, Hungary, Poland and Romania) occurred since 2007. By doing so, we seek to fill a gap in the literature on monetary policy spillovers, which is dominated by studies investigating the impact of the Federal Reserve's (Fed) monetary policy decisions and actions on emerging market economies.

We conduct a comprehensive event-study analysis to trace the impact of the ECB's nonstandard monetary policy announcements on a wide range of financial assets from CEE countries. The event-study analysis is carried out over the period 2007-2015 using daily data and controlling for, among other things, domestic and foreign monetary policy decisions and news stemming from releases of macroeconomic data in respective CEE countries. This paper covers a complete range of non-standard monetary policy measures announced by the ECB since 2007, including the recently announced expanded Asset Purchase Programme. More than seventy events, including press conferences, press releases and speeches, related to ECB's non-standard monetary policy measures are identified and systematically classified.

We find evidence of strong spillovers from the ECB's non-standard monetary policy measures to all CEE countries, in particular, on sovereign bond yields. We find that spillovers from the Securities Markets Programme (SMP) announcements were the most pronounced, while spillovers from the Outright Monetary Transactions (OMT) and the Public Sector Purchase Programme (PSPP) announcements were rather limited. Turning to the transmission channels of these spillovers, we argue that for the SMP announcements the portfolio rebalancing and the signalling channels played a key role. The OMT has impacted CEE countries indirectly, mainly through the confidence channel – by reducing the perceived redenomination risk within the euro area – without resulting in cross-border spillovers. Regarding the PSPP, we find evidence that it operated via both the confidence and the signalling channels.

"QE has several effects. [...] The portfolio rebalancing effect, namely if you buy euro-denominated assets, people who will get cash, will buy perhaps non-euro-denominated assets, and you have a portfolio rebalancing effect through that channel."

(Mario Draghi, 4 December 2014)<sup>1</sup>

## 1. Introduction

In recent years the European Central Bank (ECB) has announced and implemented a series key policy rate changes and non-standard monetary policy measures. These actions were taken in response to a number of unusual economic and financial events, with the intention of addressing a range of risks including the risk of a too prolonged period of low inflation, fragmentation that would stem from a redenomination risk and disturbances to the liquidity of certain asset markets in the euro area (ECB, 2010; 2012; 2014a). While these actions were motivated and justified on the basis of events within the euro area, this paper suggests that they were likely to have had consequences for EU countries from Central and Eastern Europe (CEE) which, despite being outside the euro area, are closely integrated with it via real and financial linkages.

The existence of close economic and financial linkages between CEE countries and the euro area has been extensively studied and is well documented in the literature.<sup>2</sup> In general, CEE countries are small and highly open economies for which euro area countries are the dominant trading partners and providers of capital through foreign direct investments and portfolio investments. In addition, banks in CEE countries are to a large extent controlled by banking groups domiciled in the euro area, which has resulted in substantial cross-border banking flows particularly prior to the global crisis. The very high degree of economic and financial integration between CEE countries and the euro area provides us with an interesting test case for studying potential monetary policy spillovers. To the best of our knowledge, to date this region has received relatively little attention in the emergent monetary policy spillovers literature.

<sup>&</sup>lt;sup>1</sup> Introductory Statement to the press conference, 4 December 2014.

<sup>&</sup>lt;sup>2</sup> See, among others, Galesi and Sgherri (2009), Backé et al. (2013), and Sun et al. (2013).

The CEE region consists of a rather heterogeneous group of countries in terms of monetary policy regimes. Several countries follow an inflation targeting strategy with either freely or managed floating exchange rates. There are also countries, which target a bilateral exchange rate vis-à-vis the euro under a currency board arrangement (Bulgaria) or follow the exchange rate jointly with a definition of price stability (Croatia). There are also several CEE countries, which have already joined the euro area (Estonia, Latvia, Lithuania, Slovakia and Slovenia). Our paper focuses on the first group of CEE countries – the inflation targeters, specifically the Czech Republic, Hungary, Poland and Romania. In line with the Trilemma hypothesis,<sup>3</sup> these countries could pursue an independent monetary policy under an open capital account and a flexible exchange rate regime.<sup>4</sup> Rey (2013) recently argued that the global financial cycle has transformed the Trilemma hypothesis into a dilemma, suggesting that an independent monetary policy is possible "…if and only if the capital account is managed directly or indirectly", as countries which are open to the free movement of capital are exposed to monetary policy spillovers from major developed economies.

Our paper investigates whether such monetary spillovers from the euro area to CEE countries following an inflation targeting monetary policy strategy occurred since 2007. This period covers two major shocks that hit European economies, namely the global financial and economic crisis and the euro area sovereign debt crisis. By doing so, we seek to fill a gap in the literature on monetary policy spillovers, which is dominated by studies investigating the impact of the Federal Reserve's (Fed) monetary policy announcements and actions on emerging market economies. Our paper builds on the contributions of this literature, as the mechanisms and results outlined in those papers provide a useful benchmark for our analysis.

We conduct a comprehensive event-study analysis to trace the impact of the ECB's monetary policy decision announcements on a wide range of financial assets from CEE countries. The event-study analysis is carried out over the period 2007-2015 using daily data and controlling, among other things, for domestic and foreign (i.e. Fed) monetary policy decisions

<sup>&</sup>lt;sup>3</sup> The *Impossible Trinity*, or the Trilemma hypothesis of international finance documented by Obstfeld et al. (2005), suggests that it is impossible to follow a fixed exchange rate, an independent monetary policy and allow the free movement of capital all at the same time.

<sup>&</sup>lt;sup>4</sup> It should be noted that the Czech National Bank intervened in the foreign exchange market in November 2013. According to the Czech National Bank (2014) "the policy objective of using the exchange rate as an additional monetary policy instrument – and therefore of using foreign exchange interventions to weaken the koruna – is the same as in the case of interest rates: to maintain price stability in the Czech economy in line with the CNB's inflation target (which has been set at 2% since 2010)." This intervention is taken into account in the empirical analysis.

and news stemming from releases of macroeconomic data in CEE countries. In this paper we cover a complete range of non-standard monetary policy measures announced by the ECB since 2007, including the recently announced expanded Asset Purchase Programme (APP). More than seventy events, including press conferences, press releases and speeches, related to the ECB's non-standard monetary policy measures are identified and systematically classified.

We find evidence of strong spillovers from the ECB's monetary policy to all countries, in particular, on bond yields and the exchange rate. Among recent ECB monetary policy decisions, we find that spillovers from the Securities Markets Programme (SMP) announcements were the most pronounced, while spillovers from the Outright Monetary Transactions (OMT) and the Public Sector Purchase Programme (PSPP) announcements were rather limited. Turning to the channels of transmission of these spillovers, we argue that for the SMP announcements both the portfolio rebalancing and the signalling channels played a key role in the transmission to CEE countries. The OMT operated via the confidence channel, reducing the perceived redenomination risk within the euro area without resulting in cross-border spillovers to CEE economies. Regarding the PSPP we find that both the confidence and the signalling channels were important in its transmission across borders.

Although the primary focus of this paper is on spillovers from the ECB's monetary policy we also review the impact of the Fed's QE and tapering announcements. Our analysis reveals that CEE countries were subject to monetary policy spillovers not only from the euro area, but also from the US. When comparing the magnitude of spillovers from monetary policy announcements we conclude that spillovers to CEE countries from ECB policies were stronger and affected more asset classes than spillovers from Fed's policies. We also find that the average impact of announcements related to tapering was much higher than for the Fed's QE announcements, which may indicate non-linear or asymmetric spillovers during the life cycle of non-standard monetary policy measures.

The remainder of the paper is organized as follows. Section 2 provides a brief review of literature on monetary policy spillovers. Section 3 presents the non-standard measures announced and largely implemented by the ECB since 2007. Section 4 discusses the main channels through which monetary policy in advanced economies spills over internationally. Section 5 discusses the event-study methodology and reports our empirical results. Section 6 concludes.

## 2. Literature review

While the effectiveness of non-standard monetary policy measures on domestic variables has been studied extensively,<sup>5</sup> the analysis of the spillover effects of such policies from advanced economies to emerging markets has only started to receive more attention in the empirical literature in recent years.

Many of the existing papers on the topic suggest that important global externalities arise from unconventional monetary policy decisions taken in advanced economies. Chinn (2013) provides evidence about their impact on exchange rates and asset prices in emerging market economies, concluding that these policies support global rebalancing by encouraging emerging market currency revaluation. Fic (2013) examines the financial market impact of unconventional monetary policy decisions adopted in major developed countries on the BRIC economies (i.e. Brazil, China, India and Russia), documenting their impact on long term bond yields, equity prices and exchange rates. Fratzscher et al. (2013) study the global spillovers of the Fed's unconventional monetary policy measures and conclude that such policies affected capital flows to emerging market economies in a pro-cyclical manner, have raised asset prices globally and weakened the US dollar. Lim et al. (2014) study the effects of quantitative easing (QE) policies in the US on gross financial inflows to developing countries, finding that QE have been transmitted internationally through liquidity, portfolio rebalancing, and confidence channels.

The analysis by Berge and Cao (2014) shows that a change in monetary policy at the zerolower bound in the United States is associated with movements in a variety of asset prices abroad. Rogers et al. (2014) observe that there are important cross-country spillovers from unconventional monetary policies in the US, the UK, the euro area and Japan among these advanced economies. They find such monetary policy spillovers to be asymmetric, as the effects of the monetary policy shocks in the US economy on asset prices in the other three economies are larger than the spillovers from these countries' policies on the US. Tillmann (2014) finds that Fed's QE policies had strong effects on financial conditions in emerging markets and played an important role in explaining capital inflows, equity price and exchange rate movements in these economies.

<sup>&</sup>lt;sup>5</sup> See for example Angelini et al. (2011); Beirne et al. (2011); Gagnon et al. (2011); Joyce et al. (2011); Krishnamurthy and Vissing-Jorgensen (2011); Peersman (2011); Abbassi and Linzert (2012); Chen et al. (2012); D'Amico et al. (2012); Glick and Leduc (2012); Baumeister and Benati (2013); Krishnamurthy et al. (2014).

Aizenman et al. (2014) evaluate the impact of the Fed's tapering announcements from 2013 on financial markets in emerging economies and conclude that countries with stronger fundamentals face a larger depreciation of exchange rate, fall in stock prices and increase of CDS spreads than countries with weaker fundamentals.<sup>6</sup> Eichengreen and Gupta (2014) show that countries with larger and deeper financial markets experienced more pressure on exchange rates, foreign reserves and equity prices following the Fed's tapering announcement from May 2013 as investors could better rebalancing their portfolios in a country with relatively large and liquid financial markets. In contrast with Aizenman et al. (2014) and Eichengreen and Gupta (2014), Mishra et al. (2014), find that countries with stronger macroeconomic fundamentals, deeper financial markets, and a tighter macroprudential policy stance in the run-up to the Fed's tapering announcements experienced smaller currency depreciations and smaller increases in government bond yields. Burns et al. (2014) find that a normalization of unconventional monetary policies and economic activity in high-income countries implies a significant slowdown in capital inflows (specifically portfolio investments) into emerging markets. Ahmed and Zlate (2014) report evidence of positive effects of unconventional U.S. monetary policy on portfolio inflows into emerging markets.

Two main points are apparent from the above review of the literature. Firstly, the vast majority of the papers in the literature conclude that advanced economy monetary policy does indeed have spillover effects on economic and financial variables in other countries. Secondly, a substantial proportion of the existing literature focuses on spillovers from the policies of the US Federal Reserve to emerging markets. This paper contributes to the existing literature by assessing the effects of ECB monetary policy announcements on four CEE countries, all of which feature an exceptional degree of trade and financial integration with the euro area and follow inflation targeting monetary policy regime with freely or managed floating exchange rate regimes.

## 3. The ECB's non-standard monetary policy measures

In response to the global financial crisis, the ECB, similarly to other central banks from major developed economies, reduced its key interest rate significantly and implemented a number of non-standard monetary policy measures which we briefly describe below.

<sup>&</sup>lt;sup>6</sup> An interpretation of these findings given by the authors is that countries with weaker fundamentals were less exposed to the inflows triggered by quantitative easing, in line with the conjecture that being closer to financial autarky provides better insulation from financial news.

Early on in the crisis and in order to support interbank money market in the euro area the ECB introduced the following measures:<sup>7</sup> (i) an unlimited provision of liquidity through fixed rate tenders with full allotment (FRTPFA), allowing banks to get unlimited access to central bank liquidity at the main refinancing rate, subject to appropriate collateral; (ii) extension of the list of eligible collateral assets for refinancing operations (COLL); (iii) extension of the maturity of long-term refinancing operations (LTRO), in order to reduce uncertainty and improve liquidity conditions for banks; (iv) liquidity provision in foreign currencies through swap lines with other central banks to enhance banks' foreign currency funding (FOR).

In May 2009, the ECB modified the existing measures and added to them by adopting a new programme – the Enhanced Credit Support (ECS). The ECS included the four types of liquidity providing operations from the initial response outlined above and introduced a programme of outright purchases of covered bonds, the so-called Covered Bond Purchase Programme (CBPP1). A complementary program was announced in November 2011 (CBPP2). The goal of these programmes was to rekindle the functioning of the covered bond market in the euro area, which constitutes an important source of banks' refinancing in the euro area.

In May 2010, the ECB introduced an additional programme specially designed to restore an appropriate monetary policy transmission mechanism following tensions in the sovereign debt markets of some euro area countries. The Securities Markets Programme (SMP) involved purchases of euro area government bonds in the secondary markets, in order to ensure depth and liquidity in those market segments that were dysfunctional. Following the speech by the ECB President M. Draghi in London in July 2012, the ECB announced an Outright Monetary Transactions (OMT) programme in September 2012. The main aim of the OMT was to remove tail risk to overcome monetary and financial fragmentation of the euro area that would stem from a redenomination risk (ECB, 2012). The OMT, in contrast to all other ECB non-standard monetary policy measures implemented since the start of the global financial crisis, has never been applied.

In July 2013 the ECB changed its monetary policy communication strategy to include a form of forward guidance (FWG) as follows: "*The Governing Council expects the key ECB interest rates to remain at present or lower levels for an extended period of time.*" (ECB, 2013).

<sup>&</sup>lt;sup>7</sup> For more details, see de Haan et al. (2012).

In June 2014 the ECB announced measures, referred to as the Credit Easing package, to enhance the functioning of the monetary policy transmission mechanism by supporting lending to the real economy. In particular, the Governing Council decided to: i) conduct a series of targeted longer-term refinancing operations (TLTRO) aimed at improving bank lending to the euro area's non-financial private sector, excluding loans to households for house purchase, over a window of two years; ii) intensify preparatory work related to outright purchases of asset-backed securities (ABSPP), whose operations started in October 2014, in parallel with a new covered bond purchase programme (CBPP3).

In January 2015 the Governing Council announced the expanded Asset Purchase Programme (APP), which adds a purchase programme for public sector securities (PSPP) to the existing private sector asset purchase programmes (CBPP3 and ABSPP) to address the risks of a too prolonged period of low inflation, In particular, under the PSPP the ECB expanded its purchases to include bonds issued by euro area central governments, agencies and the European institutions. The combined monthly asset purchases amount to 60 billion until September 2016 or until the adjustment in the path of inflation is consistent with achieving inflation rates below, but close to, 2% over the medium term.

## 4. The international transmission of monetary policy

This section discusses the major transmission channels of international monetary policy spillovers, building on the contributions of the existing literature. More specifically, what follows is an overview of the transmission channels and a discussion of how monetary policy spillovers affect prices of the selected financial assets used in our event analysis. Our financial variables of interest include the following: (i) a bilateral nominal exchange rate vis-à-vis the euro,<sup>8</sup> (ii) yields of sovereign bonds issued in local and foreign currencies, (iii) a money market interest rate, (iv) a benchmark stock market index and (v) credit-default swaps. This selection of variables is broadly comparable with Takáts and Vela (2014), who examine the impact of Fed's policies on a broader sample of emerging markets economies, including some CEE countries.

First, we elaborate on the *confidence channel*, through which monetary policy announcements, particularly those concerning non-standard measures, help to tackle market

<sup>&</sup>lt;sup>8</sup> The euro is a reference currency for all CEE countries' currencies and its weight in their respective nominal effective exchange rate ranges from 52% in Hungary to 57% in the Czech Republic and Poland.

uncertainty in the advanced economies and beyond their borders. In general, the confidence effect may influence financial assets in either direction. For instance, improved confidence in the euro area due to a monetary policy decision may induce, through expectations, capital flows reflecting both, carry trade strategies (in which higher-yield CEE assets are often targeted) and more lasting capital flows into CEE economies because of the tight trade linkages. However, the same return of confidence may also trigger reprising of risks and capital outflows from CEE countries, particularly if these are considered to be relatively safer investments during the periods of high uncertainty.<sup>9</sup> Turning to the ECB programmes, we think that it is reasonable to assert that the international transmission of the OMT announcements operated via the confidence channel. As the OMT successfully reduced the perceived risk of sovereign bonds issued by stressed euro area countries thereby increasing capital inflows into those economies (ECB, 2014b),<sup>10</sup> its announcements might have reduced the demand for CEE sovereign bonds.

Second, monetary policy spillovers may affect CEE countries through the *portfolio rebalancing channel*. Indeed, this channel may be particularly important in the case of nonstandard monetary policies as these are generally designed to operate via changes in the prices and yields of domestic assets (Chinn, 2013). This may involve an international dimension, as substitute assets could also include comparable assets in other countries. More specifically, in the standard portfolio balance model, the ECB purchases of euro area shortand long-term government bonds would reduce their yields relative to comparable CEE bonds. Investors could turn to CEE assets for higher risk-adjusted returns, inducing lower bond yields and higher asset prices following the introduction of accommodative measures by the ECB (Fic, 2013; Mohanty, 2014).<sup>11</sup> We expect that the portfolio rebalancing channel could indeed be a prominent channel of transmission for the ECB announcements of non-

<sup>&</sup>lt;sup>9</sup> Whether CEE economies could be categorised as safe havens is debatable given that such a description was more typically applied to countries such as Switzerland, Sweden and Denmark during the crisis (Habib and Stracca, 2013; IMF, 2013).

<sup>&</sup>lt;sup>10</sup> Saka et al. (2014) show that the OMT restored confidence in the euro area sovereign markets, by signalling investors the ECB's intention to take up the function of lender of last resort.

<sup>&</sup>lt;sup>11</sup> For comparison, Mohanty (2014) explains the spillover effect of the Fed's action on emerging markets' longterm interest rates as operating through the following channel. The US long-term rate affects both the global benchmark yield and risk appetite, which together determine the pricing of bonds issued by the emerging market economies in local and international markets. Thus, given the growing presence of foreign investors in emerging markets' local currency bond markets, the monetary policy of advanced economies is likely to have a larger effect on yield curves compared to the early 2000s. These phenomena are also likely to be relevant for CEE economies. Relatedly, Adam et al. (2014) highlight the substantial portfolio inflows to Poland and the increasing share of foreign investors as holders of sovereign debt in recent years.

standard monetary policies, especially those involving direct purchases of assets.<sup>12</sup> More specifically, when the ECB buys sovereign bonds issued by the governments of the stressed euro area economies, the sellers replace those bonds with securities of similar characteristics – say bank debt, resulting in increasing bank bond prices, declining yields, and thus generally more favourable refinancing conditions for banks. However, an alternative substitute asset may have been the sovereign bonds of CEE governments. As a result, purchases of stressed euro area sovereign bonds may have had an effect on the prices and yields of CEE sovereign bonds. Among other factors, the existence and magnitude of such an effect is dependent on the degree of substitutability between bonds issued by stressed euro area countries and CEE sovereign bonds.<sup>13</sup>

Third, the *signalling channel* is closely related to the portfolio rebalancing channel in that it also operates via changes in the prices and yields of assets. What is specific to the signalling channel, however, is that it changes expectations for future short-term policy rates. This could occur if a central bank makes a public commitment to maintain low interest rates into the future, often referred to as forward guidance. This could induce a reduction of long-term interest rates through the expectations hypothesis of the term structure (Chinn, 2013) and therefore results in changes in the interest rate differentials between countries across all maturities of bonds. In turn, this could then lead to spillovers in a similar fashion to that described above in the case of the portfolio rebalancing channel.

Fourth, within the CEE economies we see some evidence to support the assertion by Takáts and Vela (2014) that central banks in emerging markets directly respond to changes in advanced economy policy rates by changing their own policy rates.<sup>14</sup> As domestic policy rate changes are an explanatory variable in our event study, it might raise an issue regarding endogeneity suggesting that our estimates of the spillover effects are biased downward.

<sup>&</sup>lt;sup>12</sup> Eser and Schwaab (2013) provide a detailed assessment of the SMP. It should be noted that, in our analysis of the effect of the SMP, we analyse only the effects of announcements, not the effect of actual purchases.

<sup>&</sup>lt;sup>13</sup> Subject to a caveat, on which we elaborate below, we think that CEE sovereign bonds might have been considered by investors as more attractive than for instance banks' debt especially at the peak of euro area sovereign debt crisis, when the feedback loop between sovereigns and banks was particularly strong. The main reason why CEE bonds are not perfect substitutes for the euro area securities is FX risk which investor would need to carry, should they decide to purchase CEE bonds denominated in local currencies.

<sup>&</sup>lt;sup>14</sup> Takáts and Vela (2014) investigate whether the responsiveness of emerging market central banks to monetary policy actions in advanced economies varies between exchange rate and inflation targeting countries. Their results indicate that the correlation between US and emerging markets' policy rates is actually stronger for inflation targeting regimes than for all emerging markets taken together.

However, this is likely to relate more to conventional interest rate policy changes than to the non-standard monetary policy measures, in which case the policy response for a central bank from the CEE region seems to be less obvious. We partially address the issue of policy endogeneity through the inclusion of interaction terms controlling for occasions when the ECB and CEE central banks policies were implemented on the same day.

Fifth, in spite of the potentially endogenous response of policy rates, a major challenge facing central banks is the disruption of the conventionally understood monetary policy transmission mechanism arising from global financial integration (Hume and Sentance, 2009; Lane and Milesi-Ferretti, 2007). For instance, the opening up of financial systems and the rise in cross-border financial flows can influence domestic credit conditions through a number of channels, including via banks access to the inter-bank and money markets, as well as international bond and equity issuance (Lane and McQuade, 2014). Most notably, an increase in liquidity in the euro area following a policy action by the ECB may have relatively direct and straightforward consequences for CEE countries because of the large presence of foreignowned banks in their local banking systems, which are controlled by parent banks domiciled almost exclusively in the euro area countries. In other words, a reduction in the interbank rate available to the parent bank is also likely to be associated with a similar reduction in the cost of funding for the subsidiary.<sup>15</sup> Furthermore, this *international bank lending channel* is likely to be particularly important in CEE economies because of the dominance of banks in financial intermediation.

Bearing these transmission channels in mind, it is important to distinguish how monetary policy spillovers may vary depending on the specific design of the monetary policy instrument announced and/or implemented by the ECB.<sup>16</sup> Much of the focus of this paper is on the effects of the SMP, the OMT and the PSPP announcements, primarily because of the similar nature of the three programmes (in principle they involve direct purchases of euro area sovereign bonds) and the contrasting impact that these programmes are found to have in

<sup>&</sup>lt;sup>15</sup> It should be noted, however, that the borrowing costs of parent banks and subsidiaries are not perfectly correlated. Furthermore, many euro area banks have sought to reduce their exposure to CEE countries by substituting domestic deposits for parent funding. To some extent, this might have been a consequence of tight funding conditions in the euro area.

<sup>&</sup>lt;sup>16</sup> The importance of distinguishing between different programmes is highlighted by Fratzscher et al. (2013), who find that the effect on emerging markets of US QE1 differed substantially from that of QE2. Similarly, Lim et al. (2014) illustrate that the effect of QE on emerging markets varied across asset classes with the most notable consequences observable for portfolio debt securities.

the event study, as outlined in Section 5. Conveniently these programmes are also relatively tractable in terms of identifying the channels through which they might operate.

It should also be noted that the degree of international transmission of ECB policies is likely determined by the underlying objective of these policies. The majority of non-standard monetary policy measures aimed to improve the transmission mechanism of the monetary policy in the euro area (e.g. SMP and OMT), while other measures directly addressed the risks of a too prolonged period of low inflation in the euro area at a time when the policy rate had reached its lower bound (e.g. forward guidance, PSPP). For instance, forward guidance attempted to address the latter via the signalling channel, whereby the ECB provided assurance to market participants that the central bank intends to keep the policy rate low for a longer period than initially expected by market participants. According to ECB (2014c), this policy was intended to influence investors' expectations regarding future short-term rates and, through the expectations hypothesis of the term structure of interest rates, put downward pressure on longer-term interest rates. Similarly the PSPP entails monthly purchases of eurodenominated investment-grade securities issued by euro area governments and agencies and European institutions with the expectation that such purchases will reduce euro area bond yields across instruments, maturities and issuers (ECB, 2015) with the final objective of addressing the risks of a too prolonged period of low inflation. What is important to note is that, despite the different motivation of these policies as compared to other non-standard monetary policy measures, the potential channels of their international spillovers remain the same.

## 5. Event-study analysis

## 5.1 The methodology

Our event-study analysis, through which we assess the spillovers from the ECB's monetary policy announcements on financial assets in CEE countries, uses the ordinary least squares estimation. We estimate the following country-specific equation using daily data over the period 01:01:2007 - 27:01:2015:

$$\Delta X_{t} = \alpha + \beta_{1} \Delta VIX_{t} + \beta_{2} \Delta IR_{t}^{Dom} + \beta_{3} \Delta IR_{t}^{ECB} + \beta_{4} NSMP_{t}^{Dom} + \delta NSMP_{t}^{ECB} + \Upsilon NSMP_{t}^{US} + \beta_{5} IMF/EC_{t} + \lambda News_{t} + u_{t}$$

$$(1)$$

where  $X_t$  is our dependent financial variable of interest (i.e. exchange rate vis-à-vis the euro, stock market index, 3-month interbank rate, medium- and long-term sovereign bond yields, yields of sovereign bonds denominated in foreign currency,<sup>17</sup> 5- and 10-year CDS spread). The  $VIX_t$  is the volatility index for the euro area and we employ it to control for periods of heightened volatility in euro area financial markets, which could in turn affect financial markets in the region of our interest. The variables  $IR_t^{Dom}$  stands for the domestic central bank policy rate and  $IR_t^{ECB}$  for the ECB policy rate (see Appendix B; Table 1B). The variable  $NSMP_t^{Dom}$  is a country specific event-dummy associated with announcements of nonstandard monetary policy decisions by the domestic monetary policy authority (see Appendix B; Table 2B). The vector  $NSMP_t^{ECB}$  contains the event-dummies related to ECB announcements of non-standard policy measures (see Appendix B; Table 3B for our classification of the ECB non-standard monetary policy measures). We use two specifications of the latter variable. First, we combine all ECB announcements regarding the various nonstandard measures. Second, we build a policy-specific variable to separate the different ECB measures in line with our classification. By including the vector  $NSMP_t^{US}$  we also control for spillovers from non-standard monetary policy announcements by the Fed (see Appendix B; Table 4B). As control variables we also introduce announcements related to the financial assistance programmes by the IMF and the European Commission  $(IMF/EC_t)$  in three out of four CEE countries (see Appendix B; Table 5B).<sup>18</sup>

Following the approach adopted by Altavilla and Giannone (2014), as additional control variables we introduce news stemming from releases of macroeconomic data,  $News_t$ , which could have affected our financial variables. A similar "controlled" event-study was recently conducted by Altavilla et al. (2014) to investigate the effects of the ECB's OMT announcements on sovereign bond yields in euro area countries. Data on  $News_t$  are collected via Bloomberg and consist of expectations of market participants about all available macroeconomic variables in the respective country. The expected values are median forecasts collected up to one day before the official data release. For each of the variables (see Appendix B; Table 6B), we compute the difference between the actual value on the day of release and its expected value. The series are standardised. They can be considered as a measure of the surprise content of the most relevant macroeconomic data releases in our

<sup>&</sup>lt;sup>17</sup> Appendix B; Table 7B reports the ISIN codes of the FX bonds used in our analysis.

<sup>&</sup>lt;sup>18</sup> These announcements concern Hungary, Poland and Romania.

countries. This procedure allows us to control for possible movements in our dependent variables that are due to unexpected changes in macroeconomic variables.

The financial variables (Appendix B, Figure 1B-9B) in equation (1) are transformed as follows. The VIX volatility index, the bilateral exchange rate vis-à-vis the euro, the stock market index, the interbank interest rate and the CDS spreads are expressed as daily percentage changes, whereas bond yields are expressed as daily basis point changes.<sup>19</sup> For the monetary policy variables and the *IMF/EC*<sub>t</sub> dummies we use a one-day event window.<sup>20</sup> In other words, around the event window we compute basis point changes of the policy rates, while the non-standard monetary policy dummy variables take the value one on the day of a particular announcement and zero elsewhere. Our motivation is to analyse international spillovers, which are likely to have longer transmission lag compared to policies implemented domestically. Therefore, in our view, this justifies the use of daily data rather than higher frequency (i.e. intra-day) data sometimes employed in event-studies.

## 5.2 The results

## **5.2.1** The aggregate measure of ECB's policies

When taking an aggregate view on ECB announcements by bundling together the various non-standard measures, we find spillover effects on sovereign bond yields (see Table 1).<sup>21</sup> More specifically, the announcements of the ECB's non-standard monetary policy measures were associated with declining sovereign bond yields in the Czech Republic, Poland and Romania. Yields of sovereign bonds issued by Hungary are the only ones from the CEE region, which seem to be un-affected by the ECB's announcements. This may be attributable to the perceived higher riskiness of the country in comparison with its regional peers as is evident from the fact that Hungary has the lowest credit rating among CEE countries examined in this paper.

<sup>&</sup>lt;sup>19</sup> Specification and the data sources are reported in Appendix A.

 $<sup>^{20}</sup>$  Alternatively we use 2-day and 3-day event windows to test for robustness of our baseline specification (see Section 5.3).

<sup>&</sup>lt;sup>21</sup> Regarding standard monetary policy, the announcements of negative changes of the key ECB policy rate in our sample coincide with a decline in sovereign bond yields denominated in local currency for Hungary and Poland. We interpret this as an evidence of carry trades, or yield-searching. We note that most of domestic and ECB monetary policy decisions on key interest rates announced during our sample period were in one direction (i.e. towards a more expansionary monetary policy stance, see Appendix B; Table 1B), which might complicate a generalisation of our results.

	ECB's non-standard policies				Do	mestic non-stand	lard polici	es
	CZ	HU	PL	RO	CZ	HU	PL	RO
Exchange rate	-0.05 (0.05)	0.07 (0.07)	-0.00 (0.07)	0.02 (0.04)	0.09 (0.12)	-0.31 (0.35)	-	-
Stock market index	-0.01 (0.16)	0.15 (0.23)	-0.13 (0.16)	-0.02 (0.17)	-0.25 (0.39)	0.95 (1.13)	-	-
Interbank rate (3M)	-0.12 (0.11)	0.17* (0.10)	-0.01 (0.05)	-0.45** (0.23)	0.06 (0.26)	-0.07 (0.48)	-	-
Benchmark bond yield (2/3Y; LC)	-1.84* (1.06)	0.52 (1.39)	0.10 (0.66)	-2.21 (4.90)	-1.84 (2.54)	-8.65 (6.72)	-	-
Benchmark bond yield (10Y; LC)	-1.03* (0.56)	1.13 (1.31)	-2.36*** (0.64)	-5.38*** (2.05)	-0.52 (1.35)	-12.14* (6.30)	-	-
Bond yield (FX)	-1.15*** (0.39)	-0.64 (0.87)	-0.01 (0.78)	0.81 (0.93)	-0.17 (0.93)	-11.69*** (4.21)	-	-
CDS 5-year (USD)	0.59 (0.53)	0.50 (0.38)	0.29 (0.45)	0.14 (0.32)	0.68 (1.27)	-1.89 (1.83)	-	-
CDS 10-year (USD)	0.28 (0.48)	0.38 (0.36)	-0.32 (0.40)	-0.23 (0.31)	-0.28 (1.16)	-1.87 (1.73)	-	-

Table 1. Event-study analysis on ECB's monetary policy decisions (1-day window)

**Note:** In addition to variables noted in the equation in Section 5.1, the regressions include two interaction terms: one for domestic and ECB key policy rate changes and the one for domestic and ECB non-standard monetary policy measures. In addition, we also include a dummy variable for the exchange rate regression for the Czech Republic in order to account for a regime change since 9 November 2013 (i.e. since when the CNB intervened and declared that the 27 CZK per EUR level is the preferred one). The  $\delta$ - and  $\beta_4$  coefficients from equation (1) are reported in this table jointly with standard errors in parentheses; \*\*\* p<0.01, \*\* p<0.05, \* p<0.1 **Source:** Authors' calculations.

As all of the CEE countries included in our event analysis pursue an inflation targeting monetary policy strategy (and thus run their monetary policy independently), the ECB non-standard monetary policy should not impact the local 3-month money market rates directly, according to the policy trilemma outlined earlier. However, our aggregated control variable for ECB non-standard monetary policy announcements is statistically significant for Hungary and Romania. Interestingly, the announcements have a positive impact on the local interest rate in the former, while negative in the latter country.

As mentioned above, one spillover channel of non-standard monetary policy actions by the ECB could be through banks' liquidity (the so called international bank lending channel), as the majority of banks operating in CEE countries are owned by groups residing in the euro area. More specifically, all longer-term liquidity operations by the ECB vis-à-vis eligible counterparties in the euro area might spillover into the CEE region through the liquidity managed at the level of the banking groups. A bank operating in the CEE region could receive liquidity from its parent bank and use it to substitute for liquidity available in the local money market. This in turn would result in a lower demand for funds at the local money market and declining money market rates. Developments consistent with this channel are observed in the case of Romania, while the ECB's non-standard monetary policy announcements coincide with increasing 3-month interest rate in Hungary. In addition to the country's low credit rating, this may also reflect behaviour of parent banks domiciled in the

euro area toward their branches and subsidiaries in Hungary, particularly in response to policy measures targeting the banking sector that were introduced by Hungarian authorities during our sample period (e.g. bank levy, early repayment scheme for FX mortgages, etc.). Parent banks responded to these policies by withdrawing a significant amount of funding from their Hungarian operations.

We find that the exchange rates of CEE currencies vis-à-vis the euro seem to have been broadly insulated from the ECB announcements. Similarly, there seem to be no spillovers from the ECB's non-standard monetary policy announcements to CDS spreads. Similarly, benchmark stock market indices in CEE countries were not affected. When interpreting the latter result one should note the less developed state of equity markets compared to debt markets in CEE countries.

We also review the impact of the Fed's QE announcements, including those related to a tapering of bond purchases, on CEE countries' financial assets.<sup>22</sup> The results are reported in Table 2. Starting with the Fed's QE announcements, we find that yields of sovereign bonds denominated in foreign currencies declined in all CEE countries, except Romania. Interestingly, yields of local currency sovereign bonds increased in Hungary and Poland. This contrasts with the spillovers from the ECB's announcements, which coincide with declining sovereign bond yields irrespective of their denomination. According to the BIS (2015), if investors treat bonds denominated in different currencies as close substitutes, purchases in one market also depress yields elsewhere. The results suggest that there may be a greater degree of substitutability between local currency denominated US Treasury bonds.

Turning to the Fed's announcements related to tapering of bond purchases under its QE program, we find that almost all asset classes are impacted in at least one country, with the notable exception of the stock market index. Moreover the average impact of the Fed's tapering announcements was much higher than those of the Fed's QE, which suggests non-linear spillovers to CEE countries from the Fed's announcements.<sup>23</sup> These results contrast with Chen et al. (2014) because we find a larger impact for the Fed's tapering announcements

 $<sup>^{22}</sup>$  Due to different time zones and the timing of Fed's announcements, we investigate the behaviour of financial assets on the day following the day of announcement.

<sup>&</sup>lt;sup>23</sup> Monetary policy spillovers may be non-linear in so far as larger shocks might result in disproportionately larger spillovers, or asymmetric in a sense that negative shocks have larger spillovers than positive shocks.

than for QE announcements even after controlling for uncertainty, for which we use the VIX as a proxy. Overall, we show that CEE countries were subject to monetary policy spillover effects from both the euro area and the US.

		Fed's QE				Fed's	tapering	
	CZ	HU	PL	RO	CZ	HU	PL	RO
Exchange rate	-0.01	0.14	0.02	-0.00	0.08	0.48*	0.12	0.24
	(0.11)	(0.16)	(0.15)	(0.10)	(0.18)	(0.27)	(0.25)	(0.16)
Stock market index	0.50	-0.46	0.05	0.16	-0.06	0.16	-0.14	-0.01
	(0.36)	(0.53)	(0.35)	(0.39)	(0.60)	(0.88)	(0.59)	(0.64)
Interbank rate (3M)	0.00	0.04	-0.13	0.06	0.12	0.11	-0.08	6.46***
	(0.24)	(0.22)	(0.10)	(0.52)	(0.40)	(0.37)	(0.17)	(0.85)
Benchmark bond yield (2/3Y; LC)	-3.03	-0.40	1.79	-4.01	-6.09	13.79***	6.25**	0.60
	(2.36)	(3.15)	(1.48)	(11.11)	(3.95)	(5.20)	(2.47)	(18.32)
Benchmark bond yield (10Y; LC)	0.50	5.10*	2.39*	-3.64	7.55***	10.59**	11.09***	11.18
	(1.26)	(2.95)	(1.43)	(4.64)	(2.10)	(4.88)	(2.39)	(7.65)
Bond yield (FX)	-4.52***	-3.46*	-4.06**	-0.63	4.34***	11.06***	2.27	10.24***
	(0.87)	(1.97)	(1.76)	(1.99)	(1.45)	(3.26)	(2.94)	(3.28)
CDS 5-year (USD)	1.35	-0.27	0.19	-0.37	1.33	3.55**	2.58	2.54**
	(1.18)	(0.86)	(1.00)	(0.72)	(1.97)	(1.41)	(1.67)	(1.18)
CDS 10-year (USD)	1.18	0.40	0.76	-0.36	1.41	3.72***	2.33	2.51**
	(1.08)	(0.81)	(0.89)	(0.71)	(1.80)	(1.34)	(1.49)	(1.17)

Table 2. Event-study analysis on Fed's monetary policy decisions (1-day window)

**Note:** In addition to variables noted in the equation in Section 5.1, the regressions include two interaction terms: one for domestic and ECB key policy rate changes and the one for domestic and ECB non-standard monetary policy measures. In addition, we also include a dummy variable for the exchange rate regression for the Czech Republic in order to account for a regime change since 9 November 2013 (i.e. since when the CNB intervened and declared that the 27 CZK per EUR level is the preferred one). The  $\gamma$ -coefficients from equation (1) are reported in this table jointly with standard errors in parentheses; \*\*\* p<0.01, \*\* p<0.05, \* p<0.1 **Source:** Authors' calculations.

In order to assess the relative size of spillovers from the ECB and Fed's policies on financial assets in CEE countries we modify equation (1) by introducing each of the  $NSMP_t^{ECB}$  and  $NSMP_t^{US}$  events as a unique variable. We then compute the sum of changes on the announcement days and run an F-test for abnormal returns. For comparability, in the variable  $NSMP_t^{ECB}$  we include only those programmes that involve purchases of sovereign assets, namely the SMP, the OMT and the PSPP.

When comparing the cumulative impact of spillovers, we conclude that spillovers from the Fed's policy announcements on bond yields were of a similar magnitude to those associated with ECB announcements, but the former impacted yields of sovereign bonds denominated in foreign currencies, while the latter yields of sovereign bonds denominated in local CEE currencies.<sup>24</sup> On average, the spillovers from the Fed's QE announcements are lower given

<sup>&</sup>lt;sup>24</sup> We note that Fed's QE announcements are associated with increasing yields of sovereign bonds denominated in local currencies in Hungary and Poland. We interpret this as an endogenous result of the investors' shift towards bonds denominated in foreign currencies around Fed's announcements. The results for the Czech Republic and Romania do not exhibit the same pattern. A further investigation of potential determinants of these responses is needed to support our intuition. We note however that Fratzscher et al. (2013) find a relatively muted impact of Fed's QE2 on yields worldwide, reflecting a large portfolio rebalancing from the global bond markets into the emerging market equities.

the higher number of events as compared to the ECB's announcements. This result is in line with the greater degree of integration between the euro area and CEE economies noted above. At the same time, the magnitude of the cumulative effect of the tapering announcements is similar to that of the QE announcements despite the lower number of the former, with the largest impact is observed in the case of Hungarian bond yields. It is notable that spillovers from the Fed's tapering announcements affected more asset classes than those of Fed's QE.

## **5.2.2** The systematic measures of ECB's policies

In our baseline model a dummy variable tracking announcements of non-standard monetary policies by the ECB bundles together the various policies over our sample period. In order to account for the fact that the ECB had announced and implemented various non-standard measures with different objectives, which in turn could imply qualitatively and quantitatively different monetary policy spillovers to CEE countries, we run two additional exercises. In the first one we estimate our baseline model using rolling windows of one year over our sample period.<sup>25</sup> Figures 11B-14B in Appendix B plot our estimated coefficient for the  $NSMP_t^{ECB}$ variable over such a rolling window for all financial assets separated by country. The following eight major ECB's announcements are highlighted in the charts to ease the interpretation: 1) the initial 6-month LTROs (28/03/2008); 2) Enhanced Credit Support (CBBB and the initial 1-year LTROs; 07/05/2009); 3) SMP (10/05/2010); 4) Announcement of the active implementation of the SMP (08/08/2011); 5) 3-year LTROs (08/12/2011); 6) Draghi's "London speech" (26/07/2012); 7) TLTROs (05/06/2014); 8) Draghi's speech at the European Parliament (17/11/2014). This simple exercise confirms our assertion that spillovers from ECB's announcements have varied across the different measures announced by the ECB. In particular, the announcement of the SMP (event labelled as #3) seems to have had a significant influence on most of our financial variables, especially bond yields and CDS spreads.

<sup>&</sup>lt;sup>25</sup> Figure 10B in Appendix B plots the number of ECB announcements of non-standard monetary policies using our preferred one-year window for the rolling regression over our sample. In order to test how sensitive is our indicator to different length of rolling window we also show two additional lines corresponding to the number of events for both, the shorter and the longer window length for rolling regression ( $\pm 100$  days). Results are broadly comparable with the baseline specification using one-year window.

	ECB's sove	ECB's sovereigns purchase measures (9 events)			Fed's QE measures (14 events)				Fed's tapering (4 events)			
	CZ	HU	PL	RO	CZ	HU	PL	RO	CZ	HU	PL	RO
Exchange rate	-2.92**	-2.43	-4.36**	0.44	-0.37	1.98	-0.22	-0.14	0.48	2.90**	0.76	0.95
Stock market index	-3.75	-4.03	-0.85	-5.78	7.50	-6.52	1.20	2.17	0.39	-0.33	-2.15	0.28
Interbank rate (3M)	-2.48	-0.64	0.38	-17.04***	0.10	0.64	-1.73	0.81	0.39	0.51	-0.38	30.17***
Benchmark bond yield (2/3Y; LC)	-20.55	-60.57*	-39.40**	-82.03	-44.42	-2.88	25.44	-75.92	5.23	73.23***	32.91***	0.41
Benchmark bond yield (10Y; LC)	15.04	-73.19**	-34.69**	-100.00*	8.02	82.59**	35.61*	-32.64	35.41***	58.35***	53.06***	55.65
Bond yield (FX)	-6.99	7.03	-20.18	-32.04	-64.37***	-52.45*	-58.80**	-13.90	18.38***	54.16***	14.15	49.98***
CDS 5-year (USD)	-33.58**	-20.66**	-26.02**	-24.86***	18.72	-4.11	1.56	-4.67	6.63	18.81***	10.95	12.71**
CDS 10-year (USD)	-33.71***	-21.62**	-23.85**	-25.23***	16.42	4.51	9.96	-4.43	6.22	19.59***	10.49	12.92**

#### Table 3. Cumulative effect of ECB's and Fed's monetary policy decisions (1-day window)

Note: The table reports the sum of changes on the announcement days of ECB's and Fed's monetary policy decisions based on a 1-day event window. The ECB's sovereigns purchase measures include SMP, OMT and PSPP events. \*, \*\*, and \*\*\* denote significance of the F-test for abnormal return at 10%, 5%, and 1%, respectively. In addition to variables noted in the equation in Section 3.1, the regressions include two interaction terms: one for domestic and ECB key policy rate changes and the one for domestic and ECB non-standard monetary policy measures. In addition, we also include a dummy variable for the exchange rate regression for the Czech Republic in order to account for a regime change since 9 November 2013 (i.e. since when the CNB intervened and declared that the 27 CZK per EUR level is the preferred one).

Source: Authors' calculations.

In order to provide further evidence, we conduct a second exercise, which consists of estimating our baseline model over the whole sample with systemised  $NSMP_t^{ECB}$  variables, including comparable non-standard monetary policy measures by the ECB. The results confirm that the spillovers from ECB announcements into CEE financial assets varied across different non-standard monetary policy measures (see Tables 8B-11B, Appendix B). The events relating to the provision of long-term refinancing operations (LTRO) and unlimited provision of liquidity through fixed rate tenders with full allotment (FRTPFA), resulted in decline of bond yields in the Czech Republic and Hungary. Similarly, the announcements on liquidity provisions in foreign currency (FOR) and forward guidance (FWG) spilled over to Czech and Polish 10-year bond yields. The announcements of the remaining measures resulted in some spillover effects on financial assets.

Among the ECB non-standard monetary policy measures which involve purchases of sovereign bonds (Table 4), the SMP announcements seem to have had the most pronounced spillovers on financial assets in CEE countries. Conversely, the OMT announcements seem to induce rather limited spillovers into CEE countries. Interestingly, no statistically significant coefficient is found for events related to the recently announced PSPP.

Table 5 shows the results in cumulative terms for ECB policies which involve purchases of sovereign securities. We show that the SMP announcements are associated with significant spillovers leading to appreciation of local currencies vis-à-vis the euro (in the Czech Republic, Hungary and Poland), declining stock market indices,<sup>26</sup> bond yields<sup>27</sup> and credit-default spreads (CDS). As regards the OMT announcements, we find only weak spillover effects on the Czech, the Polish and the Romanian sovereign bond yields, while the PSPP announcements seem to have had even more limited spillovers on financial assets across CEE countries. These results remain unchanged, when we focus on the key event for each ECB measure.

<sup>&</sup>lt;sup>26</sup> While declining stock prices in response to the SMP announcements are somewhat counterintuitive, similar findings are reported in Fratzscher et al. (2014) and Georgiadis and Gräb (2015) for emerging market economies as a whole.

<sup>&</sup>lt;sup>27</sup> Bond yields in the Czech Republic are however un-affected by the SMP announcements.

		SMP				ON	ΛT		PSPP			
	CZ	HU	PL	RO	CZ	HU	PL	RO	CZ	HU	PL	RO
Exchange rate	-1.10***	-0.73*	-1.04***	0.09	-0.28	-0.30	-0.27	0.23	0.05	0.02	-0.34	-0.09
	(0.29)	(0.43)	(0.40)	(0.25)	(0.24)	(0.36)	(0.33)	(0.21)	(0.21)	(0.31)	(0.29)	(0.18)
Stock market index	-2.58***	-4.16***	-2.02**	-4.58***	-0.53	0.20	-0.11	0.72	0.78	0.87	0.78	0.38
	(0.97)	(1.41)	(0.94)	(1.04)	(0.79)	(1.16)	(0.79)	(0.85)	(0.69)	(1.01)	(0.68)	(0.74)
Interbank rate (3M)	-1.04	-0.41	-0.23	-7.12***	-0.00	0.01	0.07	-0.13	-0.00	0.05	0.17	-0.52
	(0.65)	(0.60)	(0.28)	(1.37)	(0.53)	(0.49)	(0.23)	(1.12)	(0.47)	(0.43)	(0.20)	(0.98)
Benchmark bond yield (2/3Y; LC)	-3.77	-28.15***	-12.35***	-36.85	-3.98	-0.05	-3.38	-1.11	-0.35	-0.75	-0.63	0.16
	(6.38)	(8.40)	(3.97)	(29.66)	(5.22)	(6.90)	(3.33)	(24.36)	(4.56)	(6.02)	(2.85)	(21.24)
Benchmark bond yield (10Y; LC)	1.30	-31.50***	-3.91	-49.79***	2.57	-1.15	-5.23	1.40	0.75	-1.37	-2.26	0.50
	(3.40)	(7.86)	(3.84)	(12.33)	(2.78)	(6.45)	(3.23)	(10.12)	(2.43)	(5.63)	(2.76)	(8.83)
Bond yield (FX)	0.19	-7.86	-8.24*	-3.95	-3.82**	-0.75	-1.45	-8.16*	0.64	4.52	0.32	0.23
	(2.35)	(5.27)	(4.73)	(5.32)	(1.92)	(4.33)	(3.97)	(4.35)	(1.68)	(3.77)	(3.39)	(3.79)
CDS 5-year (USD)	-13.53***	-10.95***	-12.00***	-10.27***	-2.21	-0.31	-0.81	0.01	0.76	0.76	0.71	-1.08
	(3.19)	(2.27)	(2.68)	(1.90)	(2.61)	(1.87)	(2.25)	(1.56)	(2.28)	(1.63)	(1.92)	(1.36)
CDS 10-year (USD)	-13.06***	-11.17***	-11.55***	-10.47***	-2.62	-0.45	-0.21	-0.24	0.73	0.74	0.53	-0.82
	(2.91)	(2.14)	(2.39)	(1.87)	(2.38)	(1.76)	(2.00)	(1.54)	(2.08)	(1.54)	(1.71)	(1.34)

#### Table 4. Event-study analysis on ECB's monetary policy decisions (SMP, OMT and PSPP, 1-day window)

Note: In addition to variables noted in the equation in Section 5.1, the regressions include two interaction terms: one for domestic and ECB key policy rate changes and the one for domestic and ECB non-standard monetary policy measures. In addition, we also include a dummy variable for the exchange rate regression for the Czech Republic in order to account for a regime change since 9 November 2013 (i.e. since when the CNB intervened and declared that the 27 CZK per EUR level is the preferred one). The  $\delta$ -coefficients from equation (1) for the respective ECB non-standard monetary policy measure are reported in this table jointly with standard errors in parentheses; \*\*\* p<0.01, \*\* p<0.05, \* p<0.1 Source: Authors' calculations.

		SMP (2 events)				OMT (3	8 events)			PSPP (4 e	events)	
	CZ	HU	PL	RO	CZ	HU	PL	RO	CZ	HU	PL	RO
Exchange rate	-2.29***	-1.48**	-2.16***	0.17	-0.90	-0.87	-0.83	0.68	0.24	-0.08	-1.39	-0.42
Stock market index	-5.16***	-8.56***	-4.09**	-9.20***	-1.92	0.50	-0.35	1.76	3.26	4.02	3.64	1.51
Interbank rate (3M)	-2.10	-0.87	-0.48	-15.24***	-0.42	0.05	0.20	-0.44	-0.08	0.18	0.65	-1.48
Benchmark bond yield (2/3Y; LC)	-8.27	-58.33***	-25.28***	-79.77	-11.13	-1.05	-10.66	-5.34	-1.07	-1.61	-3.70	1.67
Benchmark bond yield (10Y; LC)	2.77	-65.28***	-8.71	-105.12***	8.10	-3.94	-16.27*	0.49	4.47	-4.31	-9.85	2.69
Bond yield (FX)	0.37	-17.13	-17.49*	-8.61	-10.35*	-1.55	-4.50	-26.06**	3.35	26.57*	1.64	1.98
CDS 5-year (USD)	-28.74***	-22.96***	-25.20***	-21.49***	-7.46	-1.23	-3.51	-0.32	1.92	3.34	2.10	-3.20
CDS 10-year (USD)	-27.76***	-23.45***	-24.28***	-21.93***	-8.78	-1.65	-1.42	-1.11	2.16	3.29	1.41	-2.37
	SM	P announceme	ent (10 May 20	)10)	OMT (Draghi's speech, 26 July 2012)				PSPP announcement (22 January 2015)			
	CZ	HU	PL	RO	CZ	HU	PL	RO	CZ	HU	PL	RO
Exchange rate	-1.99***	-0.99	-1.80***	0.09	-0.30	-0.57	-0.33	-0.13	0.21	-0.46	-0.50	-0.29
Stock market index	-2.62*	-6.60***	-2.46*	-5.06***	0.14	0.36	-0.46	1.73	1.189	2.79	2.62*	0.30
Interbank rate (3M)	-1.16	-0.88	-0.48	-17.13***	0.06	-0.08	0.10	-0.27	-0.24	0.02	0.05	1.56
Benchmark bond yield (2/3Y; LC)	-11.07	-48.70***	-18.69***	-97.53**	-2.64	2.06	-6.61	-0.01	0.90	4.05	-4.79	3.81
Benchmark bond yield (10Y; LC)	3.03	-54.66***	-13.56**	-105.21***	-0.64	-1.54	-12.18**	1.40	5.86	2.76	-5.12	2.90
Bond yield (FX)	0.18	-22.13***	-19.16***	-9.45	-7.62**	-0.73	-0.52	-3.71	3.28	34.02***	1.63	3.99
CDS 5-year (USD)	-30.39***	-21.75***	-25.00***	-19.86***	-1.49	-2.06	-0.25	-1.21	-3.02	1.85	-1.83	2.83
CDS 10-year (USD)	-29.60***	-22.42***	-24.36***	-20.40***	-1.64	-2.18	-1.09	-1.30	-1.92	1.84	-1.91	2.29

#### Table 5. Cumulative effect of ECB's monetary policy decisions

**Note:** The upper part of the table reports the sum of changes on the announcement days of ECB's monetary policy decisions based on a 1-day event window. \*, \*\*, and \*\*\* denote significance of the F-test for abnormal return at 10%, 5%, and 1%, respectively. The lower part of the table reports the coefficient on the announcement day of ECB's monetary policy decisions based on a 1-day event window with \*, \*\*, and \*\*\* denoting significance of a *t*-test at 10%, 5%, and 1%, respectively. In addition to variables noted in the equation in Section 5.1, the regressions include two interaction terms: one for domestic and ECB key policy rate changes and the one for domestic and ECB non-standard monetary policy measures. In addition, we also include a dummy variable for the exchange rate regression for the Czech Republic in order to account for a regime change since 9 November 2013 (i.e. since when the CNB intervened and declared that the 27 CZK per EUR level is the preferred one). **Source:** Authors' calculations.

## 5.2.3 Channels of transmission

Focusing on the reaction of sovereign bond yields, we assume that the main channel of transmission of spillovers from ECB's announcements of non-standard monetary policy measures was the portfolio rebalancing channel.<sup>28</sup> This would imply that investors purchase CEE sovereign bonds in exchange for sovereign bonds of euro area countries targeted by the ECB's actions, which would result in lower yields of these bonds. One can also define the portfolio rebalancing channel more broadly. For instance, Klitgaard and Lucca (2015) emphasise the importance of the Lucas tree asset pricing model (Lucas, 1978) when discussing potential channels of monetary policy spillovers across borders. In such a model it is possible for asset prices to change in response to new information without any transaction taking place. This implies that spillovers can arise due to policy announcements and not only as a flow effect when the actual purchases are being made.

Of the ECB programmes involving direct purchases of sovereign assets, only the results for the SMP provide strong evidence of lower yields in response to the ECB announcements. The results for the OMT and the PSPP announcements show virtually no reaction. Although this suggests that spillovers to financial assets of the ECB announcements of the latter two programmes were limited, it cannot be excluded that CEE countries were affected via other channels (e.g. confidence, signalling channel).

We follow Fratzscher et al. (2014) when investigating the presence of the confidence channel of transmission, through which the impact of the ECB's announcements spills over into CEE countries. We conduct an event-study around the SMP, the OMT, and the PSPP events using the euro area volatility index ( $\Delta VIX_t$ ) as dependent variable. The results (Table 6), suggest that the OMT and the PSPP events are associated with declining volatility, which in turn could be interpreted as declining uncertainty and/or improving confidence in the euro area. In contrast, the SMP announcements show no impact. Given the lack of indices tracking volatility in CEE countries and building on the existent high degree of economic and financial integration of the region with the euro area and a relatively smaller size of CEE countries compared with the euro area, we believe that a general level of uncertainty in CEE

<sup>&</sup>lt;sup>28</sup> In general, identifying channels of international transmission of monetary policy empirically is challenging, especially for countries with relatively limited availability of data. One option to test for the portfolio rebalancing channel more directly would be to employ high frequency data on capital flows. We use daily funds flows into bonds and stocks from the EPFR dataset. The results (not reported in this paper) are generally inconclusive.

countries and the euro area are highly and positively correlated. This leads us to conclude that the confidence channel is likely to have played a role in the transmission of spillovers from the OMT and the PSPP announcements.

	Euro area	Euro area	inflation-	Euro	area implied for	ward
	VIX	linked s	wap rate	infla	tion-linked swap	rates
		5-year	10-year	1-year rate 4 years ahead	1-year rate 9 years ahead	5-year rate 5 years ahead
SMP (2 events)	-5.32 (-3.68)	0.02 (0.02)	0.02 (0.01)	0.05*** (0.02)	0.01 (0.02)	0.01 (0.01)
Announcement (10 May 2010)	-	0.10*** (0.02)	0.08*** (0.02)	0.11*** (0.03)	0.04 (0.03)	0.06*** (0.02)
OMT (3 events)	-10.04*** (-3.01)	0.00 (0.01)	-0.00 (0.01)	0.01 (0.01)	-0.01 (0.02)	-0.00 (0.01)
Draghi's speech (26 July 2012)	-	0.01 -0.02	0.01 -0.02	0.01 -0.03	0.03 -0.03	0.01 (0.02)
PSPP (4 events)	-5.44** (-2.61)	0.01 (0.01)	0.01 (0.01)	0.02 (0.01)	0.00 (0.01)	0.01 (0.01)
Announcement (22 January 2015)	-	0.07*** (0.02)	0.07*** (0.02)	0.11*** (0.03)	0.07** (0.03)	0.07*** (0.02)

Table 6. Testing for the confidence and the signalling channels (1-day window)

**Note:** Standard errors in parentheses; \*\*\* p<0.01, \*\* p<0.05, \* p<0.1 **Source:** Authors' calculations.

Similarly, we use selected financial variables tracking inflation expectations in the euro area to directly test for the presence of the signalling channel.<sup>29</sup> Looking at all events by respective ECB programmes, we find only weak evidence as inflation expectations only reacted to the SMP announcements (Table 6). This evidence becomes stronger if we look at the key announcements per programme. We conclude that the SMP and the PSPP announcements by providing a signal of additional monetary policy stimulus helped increase inflation expectations in the euro area.

## 5.3 Robustness checks

A general limitation associated with event-study analyses is that it is necessary to assume that financial markets are (information) efficient, i.e. the majority of the impact of ECB monetary policy measures on financial assets of CEE countries does not occur when operations are actually implemented, but when market expectations about those measures are formed. Hence, the choice of the event window length is crucial, since it involves a trade-off between keeping the interval narrow to avoid the noise produced by extraneous information, and choosing a wider window to identify potential delayed reactions of market participants.

<sup>&</sup>lt;sup>29</sup> A similar analysis, but using inflation-linked euro area government bonds, is conducted by Georgiadis and Gräb (2015).

Moreover, asset prices abroad may react more slowly to ECB monetary policy decisions compared to euro area asset prices. For these reasons, we test the robustness of the results obtained from our baseline model by extending our event window to two and three days respectively. Our main results remain the same, although we find some evidence of a delayed response of financial markets for particular asset classes, such as the exchange rate vis-à-vis the euro and bond yields.<sup>30</sup>

It could be argued that only the surprise component of the ECB announcements should be expected to have contemporaneous asset-price effects, in either the euro area or CEE economies because, if the market already foresees announcement of new policies, this should already be reflected in asset prices prior to the announcement. Therefore, as a robustness test to our baseline specification using a binary dummy variable for the ECB announcements, we construct a variable which could capture the size of monetary policy surprise stemming from ECB announcements. Following Rogers et al. (2014), we use the daily change in the Italian-German 10-year sovereign bond yield spread on the day of the announcement. As most of ECB non-standard monetary policy measures were motivated by improving the transmission mechanism of monetary policy, if deemed successful, one should observe a compression reflects the size of the surprise component stemming from the respective announcement. The estimates (see Appendix C) generally confirm our baseline results obtained using the binary dummy variable.

## 6 Concluding remarks

This paper studies the international transmission of the ECB monetary policy. More specifically, we investigate whether the ECB's announcements of non-standard monetary policy measures spilled over to non-euro area EU countries from Central and Eastern Europe. We focus on four CEE countries that follow an inflation targeting monetary policy strategy with either freely floating exchange rate, or managed floating, namely the Czech Republic, Hungary, Poland, and Romania.

We conduct a comprehensive event-study to investigate spillovers on a set of financial assets of the CEE economies. We find strong evidence of spillovers of the ECB monetary policy to CEE countries on sovereign bond yields. Turning to specific programmes we show that the

<sup>&</sup>lt;sup>30</sup> These results are not reported here for the sake of brevity, but are available upon request.

SMP announcements had substantial spillover effects, while the OMT and the PSPP announcements seem to have had significantly more limited spillover effects on CEE countries. Turning to the transmission channels of these spillovers, we argue that for the SMP announcements the portfolio rebalancing and the signalling channels played a key role. The OMT has impacted CEE countries indirectly, mainly through the confidence channel – by reducing the perceived redenomination risk within the euro area – without resulting in cross-border spillovers. Regarding the PSPP, we find evidence that it operated via both the confidence and the signalling channels.

We see two avenues for future research. The first is to quantify the macroeconomic effects of spillovers from the ECB monetary policy on the CEE region. The second is to extend our event-study analysis to other European countries, including both emerging and developed economies.

## References

- Abbassi, P., & Linzert, T. (2012). The effectiveness of monetary policy in steering money market rates during the financial crisis. *Journal of Macroeconomics*, *34*(4), 945-954.
- Adam, M., Koziński, W., & Markun, M. (2014). Changing patterns in the dependence of long-term rates between Poland and major financial centres. *BIS Papers*(78).
- Ahmed, S., & Zlate, A. (2014). Capital flows to emerging market economies: A brave new world? *Journal of International Money and Finance*, Volume 48, Part B, 221–248.
- Altavilla, C., & Giannone, D. (2014). The effectiveness of non-standard monetary policy measures: Evidence from survey data. *ECARES Working Papers*(2014-30).
- Altavilla, C., Giannone, D., & Lenza, M. (2014). The Financial and macroeconomic effects of the OMT announcements. *Working Paper Centre for Studies in Economics and Finance* (*CSEF*)(352).
- Angelini, P., Nobili, A., & Picillo, C. (2011). The interbank market after August 2007: What has changed, and why? *Journal of Money, Credit and Banking*, 43(5), 923-958.
- Azeinman, J., Binici, M., & Hutchison, M. (2014). The transmission of Federal Reserve tapering news to emerging financial markets. *NBER Working Paper*(19980).
- Backé, P., Feldkircher, M., & Slačík, T. (2013). Economic spillovers from the euro area to the CESEE region via the financial channel: A GVAR approach. *Focus on European Economic Integration*(4), 50-64.
- Bank for International Settlements (2015, June 28). 85<sup>th</sup> Annual Report, 1 April 2014–31 March 2015. Retrieved from www.bis.org
- Baumeister, C., & Benati, L. (2013). Unconventional monetary policy and the Great Recession: Estimating the impact of a compression in the yield spread at the zero lower bound. *ECB Working Paper Series*(1258).
- Beirne, J., Dalitz, L., Ejsing, J., Grothe, M., Manganelli, S., Monar, F., Sahel, B., Susec, M., Tapking, J., Vong, T. (2011). The impact of the Eurosystem's covered bond purchase programme on the primary and secondary markets. *ECB Occasional Paper Series*(122).
- Berge, T., & Cao, G. (2014). Global effects of U.S. monetary policy: Is unconventional policy different? *Economic Review*(Q I), 1-10.
- Burns, A., Kida, M., Lim, J. J., Mohapatra, S., & Stocker, M. (2014). Unconventional monetary policy normalization in high-income countries: Implications for emerging market capital flows and crisis risks. *World Bank Policy Research Working Paper*(6830).
- Chen, H., Cúrdia, V., & Ferrero, A. (2012). The macroeconomic effects of large-scale asset purchase programmes. *The Economic Journal*, *122*(564), F289-F315.

- Chen, J., Mancini-Griffoli, T., & Sahay, R. (2014). Spillovers from United States monetary policy on emerging markets: Different this time? *IMF Working Paper WP/14/240*.
- Chinn, M. D. (2013). Global spillovers and domestic monetary policy. BIS Working Papers(436).
- Czech National Bank (2014). *The exchange rate as a monetary policy instrument FAQs*. Retrieved from www.cnb.cz.
- D'Amico, S., English, W., López-Salido, D., & Nelson, E. (2012). The Federal Reserve's large-scale asset purchase programmes: Rationale and effects. *The Economic Journal*, 122(564), F415-F446.
- de Haan, J., Oosterloo, S., & Schoenmaker, D. (2012). *Financial markets and institutions: A European perspective*. Cambridge University Press.
- Eichengreen, B., & Gupta, P. (2014). Tapering talk: The impact of expectations of reduced Federal Reserve security purchases on emerging markets. *World Bank Policy Research Working Paper*(6754).
- Eser, F., & Schwaab, B. (2013). Assessing asset purchases within the ECB's Securities Markets Programme. *ECB Working Paper Series*(1587).
- European Central Bank (2010, May 10). Press Release ECB decides on measures to address severe tensions in financial markets. Retrieved from www.ecb.europa.eu.
- European Central Bank (2012, September 6). *Press Release Technical features of Outright Monetary Transactions*. Retrieved from www.ecb.europa.eu.
- European Central Bank (2013, July 4). *Introductory statement to the press conference*. Retrieved from www.ecb.europa.eu.
- European Central Bank (2014a, June 5). *Press Release ECB announces monetary policy measures to enhance the functioning of the monetary policy transmission mechanism.* Retrieved from www.ecb.europa.eu.
- European Central Bank (2014b). Monthly Bulletin, March 2014.
- European Central Bank (2014c). Monthly Bulletin, April 2014.
- European Central Bank (2015). Economic Bulletin, Issue 2 2015.
- Falagiarda, M., & Reitz, S. (2015). Announcements of ECB unconventional programs: Implications for the sovereign spreads of stressed euro area countries. *Journal of International Money and Finance*, 53(C), 276-295.
- Fic, T. (2013). The spillover effects of unconventional monetary policies in major developed countries on developing countries. *UN/DESA Working Papers*(131).
- Fratzscher, M., Lo Duca, M., & Straub, R. (2013). On the international spillovers of US Quantitative Easing. *ECB Working Paper Series*(1557).

- Fratzscher, M., Lo Duca, M., & Straub, R. (2014). ECB unconventional monetary policy actions: Market impact, international spillovers and transmission channels. *Paper presented at the 15<sup>th</sup> Jacques Polak Annual Research Conference hosted by the International Monetary Fund.*
- Gagnon, J., Raskin, M., Remache, J., & Sack, B. (2011). The financial market effects of the Federal Reserve's large-scale asset purchases. *International Journal of Central Banking*, 7(1), 3-43.
- Galesi, A., & Sgherri, S. (2009). Regional Financial Spillovers Across Europe: A Global VAR Analysis. *IMF Working Paper*(WP/09/23).
- Georgiadis, G., & Gräb, J. (2015) Global financial market impact of the announcement of the ECB's extended asset purchase programme, *Globalization and Monetary Policy Institute Working Paper* 232, Federal Reserve Bank of Dallas.
- Glick, R., & Leduc, S. (2012). Central bank announcements of asset purchases and the impact on global financial and commodity markets. *Journal of International Money and Finance, 31*(8), 2078-2101.
- Habib, M. M., & Stracca, L. (2013). Foreign investors and risk shocks: Seeking a safe haven or running for the exit? *ECB Working Paper Series*(1609).
- Hume, M., & Sentance, A. (2009). The global credit boom: Challenges for macroeconomics and policy. *Journal of International Money and Finance*, 28(8), 1426-1461.
- International Monetary Fund (2013). Nordic Regional Report Staff Report for the 2013 Cluster Consultation. *IMF Country Report*(13/274).
- Joyce, M. A., Lasaosa, A., Stevens, I., & Tong, M. (2011). The financial market impact of Quantitative Easing in the United Kingdom. *International Journal of Central Banking*, 7(3), 113-161.
- Klitgaard, T. and and Lucca, D. (2015). Do Asset Purchase Programs Push Capital Abroad? *Liberty Street Economics*, 12 August 2015, Federal Reserve Bank of New York.
- Krishnamurthy, A., & Vissing-Jorgensen, A. (2011). The effects of Quantitative Easing on interest rates: Channels and implications for policy. *NBER Working Papers*(17555).
- Krishnamurthy, A., Nagel, S., & Vissing-Jorgensen, A. (2014). ECB Policies involving government bond purchases: Impact and channels. *Unpublished Manuscript*.
- Lane, P. R., & McQuade, P. (2014). Domestic credit growth and international capital flows. *The Scandinavian Journal of Economics*, 116(1), 218-252.
- Lane, P. R., & Milesi-Ferretti, G. (2007). The external wealth of nations mark II: Revised and extended estimates of foreign assets and liabilities, 1970–2004. *Journal of International Economics*, 73(2), 223-250.
- Lim, J. J., Mohapatra, S., & Stocker, M. (2014). Tinker, taper, QE, bye? The effect of quantitative easing on financial flows to developing countries. WB Policy Research Working Paper Series(6820).
- Lucas, R. E. (1978). Asset prices in an exchange economy. Econometrica, 46(6), 1429-1445.

- Mishra, P., Moriyama, K., N'Diaye, P. M'B. P, & Nguyen, L. (2014.) Impact of Fed tapering announcements on emerging markets. *IMF Working Paper No.* 14/10.
- Mohanty, M. S. (2014). The transmission of unconventional monetary policy An overview. *BIS Papers*(78).
- Obstfeld, M., Shambaugh, J. C., & Taylor, A. M. (2005). The trilemma in history: Tradeoffs among exchange rates, monetary policies, and capital mobility. *The Review of Economics and Statistics*, 87(3), 423-438.
- Peersman, G. (2011). Macroeconomic effects of unconventional monetary policy in the euro area. *ECB Working Paper Series*(1397).
- Rey, H. (2013). Dilemma not Trilemma: The global financial cycle and monetary policy independence. *Jackson Hole Economic Symposium*.
- Rogers, J., Scotti, C., & Wright, J. H. (2014). Evaluating asset-market effects of unconventional monetary policy: A cross-country comparison. *Fed International Finance Discussion Papers*(1101).
- Saka, O., Fuertes, A., & Kalotychou, E. (2014). ECB policy and eurozone fragility: Was de grauwe right? *CEPS Working Document*(397).
- Sun, Y., Heinz, F. F., & Ho, G. (2013). Cross-country linkages in Europe: A Global VAR analysis. *IMF Working Paper*(WP/13/194).
- Takáts, E., & Vela, A. (2014). International monetary policy transmission. BIS Papers(78).
- Tillmann, P. (2014). Unconventional monetary policy shocks and the response of emerging markets. Unpublished Manuscript.

## **APPENDIX A:** The data sources

- **Spot exchange rate vis-à-vis the euro**: expressed as the number of local currency units per euro. Source: Bloomberg.
- Stock market indices. Source: Bloomberg.
- **3-month interbank rate**. Source: Thomson Reuters Datastream.
- Benchmark bond yields (2/3- and 10-year maturity). Source: Thomson Reuters Datastream.
- **Yields of bonds denominated in foreign currency**: see Tables 7B in Appendix B for the bonds used in the analysis. For Hungary and Poland we use the first principal component of the yields of different bonds. Source: Bloomberg.
- **CDS spreads (5- and 10-year)**. Source: Thomson Reuters Datastream.
- Euro VIX (volatility index). Source: Thomson Reuters Datastream.
- Macroeconomic news, as reported in Table 6B in Appendix B. Source: Bloomberg.

## **APPENDIX B: Other tables and figures**

	8 .		• ,
	Negative changes	Positive changes	Total number of changes
ECB	14	6	20
CZ	11	5	16
HU	40	9	49
PL	15	13	28
RO	27	7	34

#### Table 1B. Number of changes in key central bank policy rates (January 2007 - January 2015)

Table 2B. Domestic unconventional monetary policy measures

Date	Country	Description
27/09/2012	CZ	FX Interventions: Possible and Preferred non-standard instrument.
01/11/2012	CZ	Forward Guidance: Rates maintained until inflation pressures "rise significantly".
28/03/2013	CZ	FX Interventions: Possibility highlighted.
04/04/2013	HU	The Monetary Council launches the Funding for Growth Scheme (FGS) of the MNB.
30/04/2013	HU	Details of the FGS were approved by the Monetary Council.
01/08/2013	CZ	FX Interventions: Increased Likelihood.
07/11/2013	CZ	FX Intervention: Begin, target CZK 27/EUR.
17/12/2013	CZ	The Board decided to continue using the exchange rate as an additional instrument for easing the monetary conditions.
06/02/2014	CZ	The Board decided to continue using the exchange rate as an additional instrument for easing the monetary conditions.
27/03/2014	CZ	The Board decided to continue using the exchange rate as an additional instrument for easing the monetary conditions.
07/05/2014	CZ	The Board decided to continue using the exchange rate as an additional instrument for easing the monetary conditions.
26/06/2014	CZ	The Board decided to continue using the exchange rate as an additional instrument for easing the monetary conditions.
31/07/2014	CZ	The Board decided to continue using the exchange rate as an additional instrument for easing the monetary conditions.

### Table 3B. ECB monetary policy measures (January 2007 - January 2015)

Date	Description	Policy rate change	Туре
08/03/2007	GovC meeting	0.25	
06/06/2007	GovC meeting	0.25	
22/08/2007	Supplementary LTROs	0	LTRO
06/09/2007	GovC meeting, supplementary LTROs	0	LTRO
08/11/2007	GovC meeting, renewal of suppl. LTROs	0	LTRO
10/01/2008	GovC meeting, US dollar liquidity providing operations	0	FOR
07/02/2008	GovC meeting, renewal of two suppl. LTROs	0	LTRO
11/03/2008	The GovC decided to conduct US dollar liquidity providing operations	0	FOR
28/03/2008	The GovC decided to conduct supplementary 6-month LTROs	0	LTRO
02/05/2008	The GovC decided to enhance US dollar liquidity providing operations	0	FOR
03/07/2008	GovC meeting	0.25	
30/07/2008	The GovC decided to enhance US dollar liquidity providing operations	0.25	FOR
31/07/2008	The GovC decide to renew two LTROs	0	LTRO

04/09/2008	GovC meeting, renewal of two LTROs	0	LTRO
18/09/2008	The GovC decided to enhance US dollar liquidity providing operations	0	FOR
26/09/2008	The GovC decided to enhance US dollar liquidity providing operations	0	FOR
29/09/2008	The GovC decided to double the temporary swap lines with the Fed	0	FOR
07/10/2008	The GovC decided to enhance a LTROs and expand US dollar liquidity providing operations	0	FOR, LTRO
08/10/2008	The GovC decided to adopt a fixed rate tender procedure with full allotment	-0.5	FRTPFA
13/10/2008	The GovC decided to conduct US dollar liquidity providing operations	0	FOR
15/10/2008	The GovC decided to expand the list of assets eligible as collateral, enhance the provision of LTROs, and provide US dollar liquidity through forex swaps	0	COLL, FOR, LTRO
06/11/2008	GovC meeting	-0.5	
04/12/2008	GovC meeting	-0.75	
18/12/2008	The GovC decided that the MROs will continue to be carried out through FRTFA for as long as needed	0	FRTPFA
19/12/2008	The GovC decided to continue conducting US dollar liquidity providing operations	0	FOR
15/01/2009	GovC meeting	-0.5	
03/02/2009	The GovC decided to extend the temporary swap lines with the Fed	0	FOR
05/03/2009	GovC meeting, the GovC decided to continue the FRTFA for MROs and LTROs for as long as needed	-0.5	FRTPFA, LTRO
19/03/2009	The GovC decided to continue conducting US dollar liquidity providing operations	0	FOR
02/04/2009	GovC meeting	-0.25	
06/04/2009	The GovC decided to establish a temporary reciprocal currency arrangement with the Fed	0	FOR
07/05/2009	GovC meeting. The GovC decided to proceed with the ECS. In particular, the GovC decided to purchase euro-denominated covered bonds issued in the euro area, and to conduct liquidity providing LTROs with a maturity of one year	-0.25	CBPP, LTRO
04/06/2009	GovC meeting, The GovC decided upon the technical modalities of CBPP1	0	CBPP
25/06/2009	The GovC decided to extend the liquidity swap arrangements with the Fed	0	FOR
24/09/2009	The GovC decided to continue conducting US dollar liquidity providing operations	0	FOR
03/12/2009	GovC meeting. The GovC decided to continue conducting its MROs as FRTFA	0	FRTPFA,
04/03/2010	GovC meeting. The GovC decided to continue conducting its MROs as FRTFA for as long as needed, and to enhance the provision of LTROS	0	FRTPFA, LTRO
	The GovC decided to proceed with the SMP, to reactivate the temporary liquidity		FOR,
10/05/2010	swap lines with the Fed, to adopt a FRTPFA in the regular 3-month LTROs, and to conduct new special LTROs	0	LTRO, SMP
10/06/2010	GovC meeting. The GovC decided to adopt a FRTPFA in the regular 3-month LTROs	0	LTRO
02/09/2010	GovC meeting. The GovC decided to continue to conduct its MROs as FRTPFA for as long as necessary, and to conduct 3-month LTROs as FRTPFA	0	FRTPFA, LTRO
02/12/2010	GovC meeting. The GovC decided to continue to conduct its MROs as FRTPFA for as long as necessary, and to conduct 3-month LTROs as FRTPFA	0	FRTPFA, LTRO
17/12/2010	The ECB announced a temporary swap facility with the Bank of England	0	FOR
21/12/2010	The GovC decided to extend the liquidity swap arrangements with the Fed	0	FOR
03/03/2011	GovC meeting. The GovC decided to continue to conduct its MROs as FRTPFA for as long as necessary, and to conduct 3-month LTROs as FRTPFA	0	FRTPFA, LTRO
07/04/2011	GovC meeting	0.25	
09/06/2011	GovC meeting. The GovC decided to continue to conduct its MROs as FRTPFA for as long as necessary, and to conduct 3-month LTROs as FRTPFA	0	FRTPFA, LTRO
29/06/2011	The GovC decided to extend the liquidity swap arrangements with the Fed	0	FOR
07/07/2011	GovC meeting	0.25	
04/08/2011	GovC meeting. The GovC decided to continue to conduct its MROs as FRTPFA for as long as necessary, to conduct 3-month LTROs as FRTPFA, and to conduct	0	FRTPFA, LTRO

	a liquidity providing supplementary LTRO with a maturity of 6 months as a FRTPFA		
08/08/2011	The GovC decided to actively implement its SMP for Italy and Spain	0	SMP
25/08/2011	The GovC decided to extend the liquidity swap arrangements with the BoE	0	FOR
15/09/2011	The GovC decided to conduct 3 US dollar liquidity-providing operations in coordination with other central banks	0	FOR
06/10/2011	GovC meeting. The GovC decided to continue conducting its MROs as FRTFA for as long as needed, to conduct 3-month LTROs as FRTPFA, to conduct 2 liquidity-providing supplementary LTROs with a maturity of 12 and 13 months as FRTPFA, and to launch a new covered bond purchase program (CBPP2).	0	CBPP, FRTPFA, LTRO
03/11/2011	GovC meeting. The GovC decided upon the technical modalities of CBPP2	-0.25	CBPP
30/11/2011	The GovC decided in cooperation with other central banks the establishment of a temporary network of reciprocal swap lines	0	FOR
08/12/2011	GovC meeting. The GovC decided to conduct 2 LTROs with a maturity of 3 years and to increase collateral availability	-0.25	COLL, LTRO
21/12/2011	Results of first 3-year LTRO	0	LTRO
09/02/2012	GovC meeting. The GovC approved specific national eligibility criteria and risk control measures for the temporary acceptance in a number of countries of additional credit claims as collateral in Eurosystem credit operations	0	COLL
28/02/2012	Results of second 3-year LTRO	0	LTRO
06/06/2012	GovC meeting. The GovC decided to continue to conduct its MROs as FRTPFA for as long as necessary, and to conduct 3-month LTROs as FRTPFA	0	FRTPFA, LTRO
22/06/2012	The GovC took further measures to increase collateral availability for counterparties	0	COLL
05/07/2012	GovC meeting	-0.25	
26/07/2012	Draghi's London speech " whatever it takes"	0	OMT
02/08/2012	GovC meeting. The GovC announced that it may undertake outright open market operations of a size adequate to reach its objective	0	OMT
06/09/2012	GovC meeting. The GovC announced the technical details of OMTs and decided on additional measures to preserve collateral availability	0	COLL, OMT
12/09/2012	The GovC decided to extend the liquidity swap arrangement with the BoE	0	FOR
06/12/2012	GovC meeting. The GovC decided to continue conducting its MROs as FRTPFA for as long as necessary, and to conduct 3-month LTROs as FRTPFA	0	FRTPFA, LTRO
13/12/2012	The GovC decided to extend the liquidity swap arrangements with the Fed	0	FOR
21/02/2013	The GovC decided to publish the Eurosystem's holdings of securities acquired under the Securities Markets Programme (SMP)	0	SMP*
07/03/2013	GovC meeting	0	
22/03/2013	Collateral rule changed for some uncovered gov-guaranteed bank bonds	0	COLL
02/05/2013	GovC meeting. The GovC of the European Central Bank (ECB) has today decided to continue conducting its main refinancing operations (MROs) as fixed rate tender procedures with full allotment for as long as necessary, and at least until the end of the 6th maintenance period of 2014 on 8 July 2014. The GovC has decided to conduct the three-month longer-term refinancing operations (LTROs) as fixed rate tender procedures with full allotment.	-0.25	FRTPFA, LTRO
16/09/2013	The GovC has decided, in agreement with the Bank of England, to extend the liquidity swap arrangement with the Bank of England	0	FOR
04/07/2013	The Governing Council expects the key ECB interest rates to remain at present or lower levels for an extended period of time. (Draghi's press conference)	0	FWG
31/10/2013	ECB establishes standing swap arrangements with other central banks	0	FOR
07/11/2013	GovC meeting. The GovC decided on 7 November 2013 to continue conducting its main refinancing operations (MROs) as fixed rate tender procedures with full allotment for as long as necessary, and at least until the end of the 6th maintenance period of 2015 on 7 July 2015. Furthermore, the GovC has decided to conduct the three-month longer-term refinancing operations (LTROs) as fixed rate tender procedures with full allotment.	-0.25	FRTPFA, LTRO
09/01/2014	GovC meeting	0	
06/02/2014	GovC meeting	0	
06/03/2014	GovC meeting	0	

03/04/2014	GovC meeting	0	
08/05/2014	GovC meeting. "The Governing Council is comfortable with acting next time"* (Draghi's press conference)	0	
05/06/2014	GovC meeting. The GovC furthermore decided to conduct a series of targeted longer-term refinancing operations (TLTROs) aimed at improving bank lending to the euro area non-financial private sector, excluding loans to households for house purchase. The GovC decided to continue conducting the Eurosystem's main and three-month longer-term refinancing operations as fixed rate tender procedures with full allotment for as long as necessary. The GovC decided to extend the existing eligibility of additional assets as collateral, notably under the additional credit claims framework, at least until September 2018, and to intensify preparatory work related to outright purchases of asset-backed securities (ABS).	-0.10	LTRO, FRTPFA, COLL, TLTRO, ABSPP
17/06/2014	The GovC decided to continue offering seven-day US dollar liquidity-providing operations	0	FOR
03/07/2014	GovC meeting. The GovC decided on further technical details for the series of targeted longer-term refinancing operations (TLTROs)	0	TLTRO
29/07/2014	ECB publishes legal act relating to targeted longer-term refinancing operations	0	TLTRO
07/08/2014	GovC meeting	0	
04/09/2014	GovC meeting. The GovC decided to purchase a broad portfolio of simple and transparent asset-backed securities (ABSs) with underlying assets consisting of claims against the euro area non-financial private sector under an ABS purchase programme (ABSPP). The GovC also decided that the Eurosystem would purchase a broad portfolio of euro-denominated covered bonds issued by MFIs domiciled in the euro area under a new covered bond purchase programme (CBPP3). The interventions will start in October 2014.	0.10	CBPP, ABSPP
18/09/2014	The ECB allots €82.6 billion in first targeted longer-term refinancing operation	0	TLTRO*
02/10/2014	The ECB announces operational details of asset-backed securities and covered bond purchase programmes	0	CBPP, ABSPP
06/11/2014	GovC meeting	0	
17/11/2014	"The GovC is unanimous in its commitment to using additional unconventional instruments [] Unconventional measures might entail the purchase of a variety of assets, one of which is sovereign bonds." (M. Draghi, speech at the EP)	0	PSPP
26/11/2014	"[] we will have to consider buying other assets, including sovereign bonds in the secondary market []" (V. Constâncio, London)	0	PSPP
04/12/2014	"Evidently we are convinced that a QE programme which could include sovereign bonds falls within our mandate." (Draghi's press conference)	0	PSPP
22/01/2015	GovC meeting. The GovC announced the expanded asset purchase programme. The GovC decided today that the interest rate for the remaining six TLTROs would be equal to the rate on the Eurosystem's MROs prevailing at the time when each TLTRO is conducted.	0	PSPP, TLTRO

**Notes:** The list of events is constructed extending the one in Falagiarda and Reitz (2015). In particular, it includes events related to press conferences, press releases and speeches, and reported in the ECB media website (https://www.ecb.europa.eu/press/html/index.en.html). GovC stands for Governing Council. Events with \* are omitted from the event study.

Table 4B. US	5 unconventional	monetary pol	icv announcements
--------------	------------------	--------------	-------------------

Date	Description	Туре
25/11/2008	Fed Announces Purchases of MBS and Agency Bonds	QE
01/12/2008	Bernanke states Treasuries may be purchased	QE
16/12/2008	FOMC Meeting: FFTR decreased to 0-0.25%	QE
28/01/2009	FOMC Meeting, Large Scale Asset Purchase (LSAP) announcement	QE
18/03/2009	FOMC Meeting, LSAP	QE
10/08/2010	FOMC Meeting, LSAP	QE
27/08/2010	Bernanke Speech at Jackson Hole	QE
21/09/2010	FOMC Meeting, LSAP	QE
15/10/2010	Bernanke Speech at Boston Fed	QE

	03/11/2010	FOMC Meeting, LSAP	QE
	26/08/2011	Bernanke Speech at Jackson Hole	QE
	21/09/2011	FOMC Meeting, LSAP	QE
	20/06/2012	FOMC Meeting	QE
	13/09/2012	FOMC Meeting, LSAP	QE
	22/05/2013	Bernanke Testimony, Tapering Announcement	Tapering
	19/06/2013	FOMC Meeting, Tapering	Tapering
	18/12/2013	FOMC Meeting, Tapering	Tapering
_	29/01/2014	FOMC Meeting, Tapering	Tapering
	11/02/2014	Yellen Testimony	Tapering

**Notes:** QE events are taken from Rogers et al. (2014). Due to time zone differences, all events are used in the regression as happening the day after the announcement.

## Table 5B. IMF/EC programs events

Date	Country	Description
13/10/2008	HU	Statement by IMF Managing Director Strauss-Kahn on Hungary
22/10/2008	HU	IMF in Talks on Loans to Countries Hit by Financial Crisis + IMF and Hungary Agree on Policies to be Supported by International Community
28/10/2008	HU	IMF Announces Staff-Level Agreement with Hungary on 12.5 Billion Euro Loan (US\$15.7 billion); European Union, World Bank to Lend, Too
30/10/2008	HU	Commission proposes financial assistance to Hungary and an increase in overall BoP loans ceiling
04/11/2008	HU	The Council approves the grant of a loan to Hungary to support its medium term balance of payments
06/11/2008	HU	IMF Executive Board approves 12.3 Billion Euro Stand-By Arrangement for Hungary
17/11/2008	HU	Request for Stand-By Arrangement-Staff Report; Staff Supplement; and Press Release on the Executive Board Discussion
19/11/2008	HU	EU provides €6.5 billion Community financial assistance to Hungary.
21/11/2008	HU	Transcript of a Press Conference on the Executive Board approval of a Stand-by Arrangement for Hungary
15/12/2008	HU	Statement by an IMF Staff Mission to Hungary
09/01/2009	HU	IMF Managing Director Dominique Strauss-Kahn to Visit Hungary
23/01/2009	HU	IMF Stand-By Arrangement - Interim Review Under the Emergency Financing Mechanism
16/02/2009	HU	IMF Announces Staff-Level Agreement with Hungary on First Review of Stand-By Arrangement
11/03/2009	HU	Signature of the Supplemental MoU.
25/03/2009	RO	EU intends to provide medium-term financial assistance to Romania of up to €5 billion, IMF Outlines Plan to Lend \$17.5 Billion to Romania
26/03/2009	RO	IMF Financial Sector Coordination Meeting on Romania, Concluding Statement by Participating Banks
14/04/2009	PL	Poland seeks Flexible Credit Line
21/04/2009	RO	Commission asks Council to provide a medium-term loan to Romania as part of coordinated multilateral financial assistance
24/04/2009	RO	IMF Letter of Intent and Technical Memorandum of Understanding, April 24, 2009
04/05/2009	RO	IMF Executive Board approves €12.9 Billion Stand-By Arrangement for Romania
06/05/2009	PL	IMF grants Flexible Credit Line
18/05/2009	HU	IMF Mission to Hungary Reaches Staff-Level Agreement on Second Review of Stand-By Arrangement
20/05/2009	HU	Joint IMF, EC Press Release on the European Banking Group Coordination Meeting for Hungary.
20/05/2009	RO	Joint IMF, EC Press Release on the European Banking Group Coordination Meeting for Romania

10/06/2009	HU	Signature of the Second Supplemental MoU.
10/06/2009	RO	IMF Romania Request for Stand-By Arrangement - Staff Report; Staff Supplements; and Press Release on the Executive Board Discussion
23/06/2009	HU	IMF Completes Second Review Under Stand-By Arrangement with Hungary and approves €1.4 Billion Disbursement
23/06/2009	RO	Commission and Romania sign Memorandum of Understanding (MoU) on € billion balance-of-payments loan
30/06/2009	HU	Hungary: Second Review Under the Stand-By Arrangement, Request for Waiver of Non- observance of Performance Criterion, and Request for Modification of Performance Criteria
08/09/2009	RO	Letter of Intent and Technical Memorandum of Understanding
16/09/2009	HU	Hungary Letter of Intent and Technical Memorandum of Understanding, September 16, 2009
21/09/2009	RO	IMF Completes First Review Under Stand-By Arrangement with Romania and approves €1.85 Billion Disbursement
25/09/2009	HU	Press Release: IMF Executive Board Completes Third Review Under Hungary's Stand-By Arrangement, Extends the Arrangement, and approves €3.7 Million Disbursement
08/10/2009	RO	IMF Romania: First Review Under the Stand-By Arrangement, Request for Waiver of Non- observance of Performance Criterion, and Request for Modification and Establishment of Performance Criteria.
19/10/2009	HU	IMF Third Review Under the Stand-By Arrangement, Requests for Extension of the Arrangement.
06/11/2009	RO	IMF Statement by the IMF Mission in Romania
16/11/2009	HU	Commission concludes third review of the EU balance-of-payments assistance for Hungary.
18/11/2009	RO	European Banking Coordination Initiative Meeting for Romania, "Parent Banks Reaffirm Commitment to Romania". + IMF Statement at the Conclusion of a Staff Visit to Romania
19/11/2009	HU	European Commission and IMF Welcome Reaffirmed Commitments of the Largest Foreign Banks in Hungary
04/12/2009	HU	IMF Letter of Intent and Technical Memorandum of Understanding, December 04, 2009
22/12/2009	HU	Fourth Review Under the Stand-By Arrangement, and Request for Modification of Performance Criteria
15/01/2010	HU	Signature of the Third Supplemental MoU.
28/01/2010	RO	EU Balance of payments programme for Romania
05/02/2010	RO	Letter of Intent and Technical Memorandum of Understanding
15/02/2010	HU	European Commission concluded its fourth review of the EU medium-term financial assistance to Hungary.
19/02/2010	RO	IMF Completes Second and Third Review Under Stand-By Arrangement with Romania and approves US\$3.32 Billion Disbursement
25/02/2010	RO	Signature of Supplemental Memorandum of Understanding
26/02/2010	RO	Financial Sector Stability Assessment
26/03/2010	HU	IMF Fifth Review Under the Stand-By Arrangement, and Request for Modification of Performance Criterion
30/03/2010	RO	Statement by IMF Managing Director Dominique Strauss-Kahn at the Conclusion of his Visit to Romania
10/05/2010	RO	Statement after joint mission to Romania
10/06/2010	HU	IMF statement on Hungary + Commission visit to Hungary for informal discussions.
16/06/2010	RO	Letter of Intent, Supplementary Letter of Intent, and Technical Memorandum of Understanding
25/06/2010	RO	Statement by IMF Mission Chief for Romania on the Ruling of the Constitutional Court
02/07/2010	PL	IMF renews Poland's Flexible Credit Line
02/07/2010	RO	IMF Completes Fourth Review Under Stand-By Arrangement with Romania and approves US\$1.146 Billion Disbursement
17/07/2010	HU	Commission postpones conclusion of fifth review on the EU's balance of payments assistance to Hungary
22/07/2010	HU	European Commission and IMF meet with the banking community active in Hungary
22/07/2010	RO	Commission and IMF welcome the reaffirmed support of parent banks to their Romanian affiliates
04/08/2010	RO	Commission concludes that conditions for third disbursement of the EU BoP assistance of €1.2 bn have been met

24.09/2010         RO         IMF Completes Fifth Review Under Stand-By Arrangement with Romania and approves 68/40 Million Disbursement           3009/2010         RO         Disservance of Performance Criterion, and Requests for Modification and Establishment of Performance Criteria           3011/2010         HU         European Commission Sulf Statement after a joint mission with the IMF and the World Bank to Romania           22/12/2010         RO         European Commission Sulf Statement after a joint mission with the IMF and the World Bank to Romania           22/12/2010         RO         Euter of Tinent, and Technical Memorandum of Understanding           07/01/2011         RO         Barter of Tinent, and Technical Memorandum of Understanding           07/01/2011         PL         IMF Completes Sixth Review Under Stand-By Arrangement with Romania and approves 69/045 Million Disbursement           20/02/2011         PL         IMF reases Poland's Flexible Credit Line           03/02/2011         RO         European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania           10/03/2011         RO         RO         European Commission Staff Statement after a joint mission of the previous jink.           17/03/2011         RO         European Commission and IMF welcome the continued support by the parent banks of the largest foreign owned banks for Romania.           18/03/2011         RO         Ed. billion Disueacones Row 635 Billion P	09/09/2010	RO	IMF Letter of Intent, and Technical Memorandum of Understanding
IMF Finih Review Under the Stand-By Arrangement, and Requests for Waver of Non- Solvey2010         Non- Performance Criteria           03/11/2010         HU         Furopean Commission Bola sistance agreement expiration.           03/11/2010         RO         European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania           22/12/2010         RO         Euter of Intent, and Technical Memorandum of Understanding           07/01/2011         RO         Butter of Intent, and Technical Memorandum of Understanding           07/01/2011         PL         IMF Completes Sixth Review Under Stand-By Arrangement with Romania and approves 60/04/S Million Disbursement           03/02/2011         PL         IMF reamounce Staff Statement after a joint mission with the IMF and the World Bank to Romania           03/02/2011         RO         European Commission Staff Statement with Romania on New 61.6 billion Precautionary Arrangement           03/02/2011         RO         European Commission and IMF welcome the continued support by the parent banks of the Ingrest foreign owned banks for Romania.           17/03/2011         RO         Ed 6 billion bond issued to assist Ireland and Romania Choose translations of the previous Ink           18/03/2011         RO         Ed 6 billion bond issued to ressist for Abilion Precautionary Stand-By Arrangement for Romania. Completes Seventh and Final Review Under the Current Stand-By Arrangement and Approxes Ed billion Dishuremement           01/04/201	24/09/2010	RO	IMF Completes Fifth Review Under Stand-By Arrangement with Romania and approves €884.0 Million Disbursement
03/11/2010         HU         European Commission BoP assistance agreement expiration.           03/11/2010         RO         European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania           22/12/2010         RO         Letter of Intent, and Technical Memorandum of Understanding           07/01/2011         RO         Diff Completes Sixth Review Under Stand-By Arrangement with Romania and approves 60/4.8 Million Disbursement           21/01/2011         PL         IMF renews Poland's Flexible Credit Line           03/02/2011         HU         IMF renews Poland's Flexible Credit Line           03/02/2011         RO         European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania.           10/03/2011         RO         European Commission and MF welcome the continued support by the Current Stand-By Arrangement and on the Seventh and Final Review of the Current Stand-By Arrangement ink           17/03/2011         RO         European Commission and IMF welcome the continued support by the parent banks of the largest foreign-owned banks for Romania           18/03/2011         RO         European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania.           25/03/2011         RO         European Commission and IMG welcome the continued support by the parent banks of the largest foreign-owned banks for Romania           10/04/2011         RO         European Commission	30/09/2010	RO	IMF Fifth Review Under the Stand-By Arrangement, and Requests for Waiver of Non- observance of Performance Criterion, and Request for Modification and Establishment of Performance Criteria
03/11/2010         RO         European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romanaia           22/12/2010         RO         Letter of Intent, and Technical Memorandum of Understanding           07.01/2011         RO         IMF Completes Sixth Review Under Stand-By Arrangement with Romania and approves 6004.8 Million Disbursement           21/01/2011         PL         IMF Staff Report for the 2010 Article IV Consultation and Proposal for Post-Program Monitoring           08/02/2011         RO         European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania.           10/03/2011         RO         European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania.           17/03/2011         RO         European Commission and IMF welcome the continued support by the precautionary Stand-By Arrangement           17/03/2011         RO         European Commission and IMF welcome the continued support by the parent banks of the largest foreign-owned banks for Romania           18/03/2011         RO         European Commission and IMF welcome the continued support by the parent banks of the largest foreign-owned banks for Romania           10/04/2011         RO         Romania-Seventh Review Under the Stand-By Arrangement for Romania-Seventh Review Under the Stand-By Arrangement and approves (B illion Dishursement           0/04/2011         HU         Furetrefact deatis on deficit cuts <t< td=""><td>03/11/2010</td><td>HU</td><td>European Commission BoP assistance agreement expiration.</td></t<>	03/11/2010	HU	European Commission BoP assistance agreement expiration.
22/12/2010         RO         Letter of Intent, and Technical Memorandum of Understanding           07/01/2011         RO         IMF Completes Sixth Review Under Stand-By Arrangement with Romania and approves 69(4):8 Million Disbursement           21/01/2011         PL         IMF rearews Poland's Flexible Credit Line           03/02/2011         HU         IMF Staff Report for the 2010 Article IV Consultation and Proposal for Post-Program Monitoring           08/02/2011         RO         European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania.           10/03/2011         RO         Stand-By Arrangement and on the Seventh and Final Review of the Current Stand-By Arrangement for Arangement on on the Seventh and Final Review of the Previous fink           17/03/2011         RO         European Commission Stoff Statement and Request for Arangement, Cancellation of Current Stand-By Arrangement and approves Rev 63: Billion Precautionary Stand-By Arrangement for Romania. Completes Seventh and Final Review Under the Current Stand-By Arrangement and approves C Billion Disbursement           01.04/2011         RO         Romania-Seventh Review Under the Stand-By Arrangement, Cancellation of Current Stand-By Arrangement, and Request for a New Stand-By Arrangement and approves C Billion Disbursement           01.04/2011         RO         Romania-Seventh Review Under the Stand-By Arrangement and calls for forther devisits and approves C Billion Disbursement           09.05/2011         RO         European Commission Staff Statement after a joint	03/11/2010	RO	European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania
07.01/2011         RO         IMF Completes Sixth Review Under Stand-By Arrangement with Romania and approves 6904.5 Million Disbursement           21.01/2011         PL         IMF renews Poland's Flexible Credit Line           03.02/2011         HU         IMF Staff Report for the 2010 Article IV Consultation and Proposal for Post-Program Monitoring           08.02/2011         RO         European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania.           08.02/2011         RO         European Commission Staff Statement and Final Review of the Current Stand-By Arrangement and on the Seventh and Final Review of the Current Stand-By Arrangement           17.03/2011         RO         E4.6 billion bond issued to assist Ireland and Romania Choose translations of the previous link           18.03/2011         RO         Earopean Commission and IMF welcome the continued support by the parent banks of the largest foreign-owned bank's for Romania           25.03/2011         RO         Romania-Seventh Review Under the Stand-By Arrangement, Cancellation of Current Stand-By Arrangement, and Request for a New Stand-By Arrangement           01.04/2011         RO         Romania-Seventh Review Under the Stand-By Arrangement (Cancellation of Current Stand-By Arrangement, and Request for a New Stand-By Arrangement and calls for further details on deficit cuts           11.04/2011         HU         Hu Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania           09.05/2011         <	22/12/2010	RO	Letter of Intent, and Technical Memorandum of Understanding
21.01/2011         PL         IMF renews Poland's Flexible Credit Line           03.02/2011         HU         IMF Staff Report for the 2010 Article IV Consultation and Proposal for Post-Program Monitoring           08.02/2011         RO         European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania.           1003/2011         RO         European Commission Staff Statement and Final Review of the Current Stand-By Arrangement           17.03/2011         RO         64.6 billion hood issued to assist Ireland and Romania Choose translations of the previous link           18.03/2011         RO         European Commission and IMF velcome the continued support by the parent banks of the largest foreign-owned banks for Romania.           18.03/2011         RO         European Commission and IMF velcome the continued support by the parent banks of the largest foreign-owned banks for Romania.           18.03/2011         RO         Romania. Completes Sevent and Final Review Under the Current Stand-By Arrangement and approves QE Billion Disbursement.           01.04/2011         RO         Romania-Seventh Review Under the Cand-By Arrangement of further details on deficit cuts           11.04/2011         HU         Gommission Staff Statement after a joint mission with the IMF and the World Bank to Romania           09.05/2011         RO         European Commission Staff Statement after a 2008 Stand-By Arrangement with Romania           09.06/2011         RO	07/01/2011	RO	IMF Completes Sixth Review Under Stand-By Arrangement with Romania and approves €904.8 Million Disbursement
03/02/2011         HU         IMF Staff Report for the 2010 Article IV Consultation and Proposal for Post-Program Monitoring           08/02/2011         RO         European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania.           10/03/2011         RO         IMF Annouces Staff Level Agreement with Romania on New €3.6 billion Precautionary Stand-By Arrangement and on the Seventh and Final Review of the Current Stand-By Arrangement           17/03/2011         RO         €4.6 billion bond issued to assist Ireland and Romania Choose translations of the previous link           18/03/2011         RO         European Commission and IMF welcome the continued support by the parent banks of the largest foreign-owned hanks for Romania.           10/04/2011         RO         European Commission and IMF welcome the continued support by the parent banks of the largest foreign-owned hank for Romania.           01/04/2011         RO         Romania-Seventh Review Under the Stand-By Arrangement, Cancellation of Current Stand-By Arrangement, and Agproves C4 Billion Disbursement           08/04/2011         HU         Commission's mission to Hungary welcomes commitment to structural reform and calls for further details on deficit cuts           11/04/2011         HU         Hungary—Post-Program Monitoring Discussions           09/05/2011         RO         European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania           09/06/2011         RO         IMF Executive	21/01/2011	PL	IMF renews Poland's Flexible Credit Line
08/02/2011         RO         European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania.           10/03/2011         RO         Statababy Arrangement and on the Seventh and Final Review of the Current Stand-By Arrangement           17/03/2011         RO         €4.6 billion bond issued to assist Ireland and Romania Choose translations of the previous link           18/03/2011         RO         European Commission and IMF welcome the continued support by the parent banks of the largest foreign-owned banks for Romania           25/03/2011         RO         European Commission and IMF welcome the continued support by the parent banks of the largest foreign-owned banks for Romania           01/04/2011         RO         European Commission and IMF welcome the continued support by the parent banks of the largest foreign-owned banks for Romania           01/04/2011         RO         Romania-Seventh Review Under the Stand-By Arrangement, Cancellation of Current Stand-By Arrangement, and Request for a New Stand-By Arrangement           08/04/2011         HU         Commission's mission to Hungary welcomes commitment to structural reform and calls for further details on deficit cuts           09/05/2011         RO         European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania           09/06/2011         RO         IMF Executive Board Concludes First Post-Program Monitoring Discussions and the Ex- Post Evaluation of Exceptional Access under the 2008 Stand-By Arrangement with Hungary	03/02/2011	HU	IMF Staff Report for the 2010 Article IV Consultation and Proposal for Post-Program Monitoring
IMF Announces Staff Level Agreement with Romania on New S6. billion Precautionary Stand-By Arrangement           17/03/2011         RO         El.6 billion bond issued to assist Ireland and Romania Choose translations of the previous link           18/03/2011         RO         European Commission and IMF welcome the continued support by the parent banks of the largest foreign-owned banks for Romania           18/03/2011         RO         European Commission and IMF welcome the continued support by the parent banks of the largest foreign-owned banks for Romania.           01/04/2011         RO         Romania-Seventh Review Under the Stand-By Arrangement, Cancellation of Current stand-By Arrangement, and Request for a New Stand-By Arrangement           01/04/2011         RO         Romania-Seventh Review Under the Stand-By Arrangement           08/04/2011         HU         Commission's mission to Hungary welcomes commitment to structural reform and calls for further details on deficit cuts.           11/04/2011         HU         Hungary—Post-Program Monitoring Discussions           09/05/2011         RO         IMF Executive Board Concludes First Post-Program Monitoring Discussions and the Ex- 15/06/2011           10/104/2011         RO         IMF Executive Board Concludes First Post-Program Monitoring Discussions and the Ex- 15/06/2011           11/04/2011         RO         IMF Executive Board Concludes First Post-Program Monitoring Discussions and the Ex- 15/06/2011           11/04/2011         RO         I	08/02/2011	RO	European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania.
17/03/2011         RO         64.6 billion bond issued to assist Ireland and Romania Choose translations of the previous link           18/03/2011         RO         European Commission and IMF welcome the continued support by the parent banks of the largest foreign-owned banks for Romania           1MF Executive Board approves New 63.5 Billion Precautionary Stand-By Arrangement and approves 64 Billion Disbursement           01/04/2011         RO         Romania, Completes Seventh and Final Review Under the Current Stand-By Arrangement and approves 64 Billion Disbursement           08/04/2011         HU         Romania-Seventh Review Under the Stand-By Arrangement           08/04/2011         HU         Commission's mission to Hungary welcomes commitment to structural reform and calls for further details on deficit cuts           11/04/2011         HU         Gummania         European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania           09/05/2011         RO         IMF Executive Board Concludes First Post-Program Monitoring Discussions and the Ex-Post Evaluation of Exceptional Access under the 2008 Stand-By Arrangement with Hungary           27/06/2011         RO         IMF Completes First Review Under Precautionary Stand-By Arrangement with Romania           29/06/2011         RO         Statement by the EC and IMF on the Review Mission to Romania           14/09/2011         RO         IMF Completes Second Review Under Stand-By Arrangement with Romania           29/06/20	10/03/2011	RO	IMF Announces Staff Level Agreement with Romania on New €3.6 billion Precautionary Stand-By Arrangement and on the Seventh and Final Review of the Current Stand-By Arrangement
18/03/2011         RO         European Commission and IMF welcome the continued support by the parent banks of the largest foreign-owned banks for Romania           25/03/2011         RO         Romania, Completes Seventh and Final Review Under the Current Stand-By Arrangement and approves (6 Billion Disbursement)           01/04/2011         RO         Romania-Seventh Review Under the Stand-By Arrangement, Cancellation of Current Stand-By Arrangement, Stand-By Arrangement, Cancellation of Current Genomission's mission to Hungary welcomes commitment to structural reform and calls for further details on deficit cuts           11/04/2011         HU         Hungary—Post-Program Monitoring Discussions           09/05/2011         RO         European Commission atd Technical Memorandum of Understanding           09/06/2011         RO         IMF Ectecutive Board Concludes First Post-Program Monitoring Discussions and the Expose/2011           11/04/2011         HU         Post Evaluation of Exceptional Access under the 2008 Stand-By Arrangement with Hungary           27/06/2011         RO         IMF Ectecutive Board Concludes First Post-Program Monitoring Discussions and the Expose/2011 <td< td=""><td>17/03/2011</td><td>RO</td><td>€4.6 billion bond issued to assist Ireland and Romania Choose translations of the previous link</td></td<>	17/03/2011	RO	€4.6 billion bond issued to assist Ireland and Romania Choose translations of the previous link
IMF Executive Board approves New 63.5 Billion Precautionary Stand-By Arrangement for Romania, Completes Seventh and Final Review Under the Current Stand-By Arrangement and approves €1 Billion Disbursement           01/04/2011         RO         Romania, Seventh Review Under the Stand-By Arrangement, Cancellation of Current Stand-By Arrangement, and Request for a New Stand-By Arrangement           08/04/2011         HU         Commission's mission to Hungary welcomes commitment to structural reform and calls for further details on deficit cuts           11/04/2011         HU         Hungary—Post-Program Monitoring Discussions           09/05/2011         RO         European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania           09/06/2011         RO         IMF Letter of Intent and Technical Memorandum of Understanding           15/06/2011         RO         IMF Executive Board Concludes First Post-Program Monitoring Discussions and the Ex- Post Evaluation of Exceptional Access under the 2008 Stand-By Arrangement with Hungary           27/06/2011         RO         IMF Completes First Review Under Precautionary Stand-By Arrangement with Romania and approves €481 Million Disbursement           29/06/2011         RO         European Commission and Romania sign Memorandum of Understanding (MoU) on EUR 1.4 billion pre-cautionary blance-of-payments assistance           01/08/2011         RO         IMF Completes Second Review Under Stand-By Arrangement with Romania           14/09/2011         RO         IMF Complete	18/03/2011	RO	European Commission and IMF welcome the continued support by the parent banks of the largest foreign-owned banks for Romania
01/04/2011         RO         Romania-Seventh Review Under the Stand-By Arrangement, Cancellation of Current Stand-By Arrangement, and Request for a New Stand-By Arrangement           08/04/2011         HU         Commission's mission to Hungary welcomes commitment to structural reform and calls for further details on deficit cuts           11/04/2011         HU         Hungary—Post-Program Monitoring Discussions           09/05/2011         RO         European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania           09/06/2011         RO         IMF Letter of Intent and Technical Memorandum of Understanding           11/04/2011         HU         Post-Program Monitoring Discussions and the Exployed to Reveal t	25/03/2011	RO	IMF Executive Board approves New €3.5 Billion Precautionary Stand-By Arrangement for Romania, Completes Seventh and Final Review Under the Current Stand-By Arrangement and approves €1 Billion Disbursement
08/04/2011         HU         Commission's mission to Hungary welcomes commitment to structural reform and calls for further details on deficit cuts           11/04/2011         HU         Hungary—Post-Program Monitoring Discussions           09/05/2011         RO         European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania           09/06/2011         RO         IMF Letter of Intent and Technical Memorandum of Understanding           15/06/2011         RO         IMF Executive Board Concludes First Post-Program Monitoring Discussions and the Ex- Post Evaluation of Exceptional Access under the 2008 Stand-By Arrangement with Hungary           27/06/2011         RO         IMF Completes First Review Under Precautionary Stand-By Arrangement with Romania and approves €481 Million Disbursement           29/06/2011         RO         EUR 1.4 billion pre-cautionary balance-of-payments assistance           01/08/2011         RO         Statement by the EC and IMF on the Review Mission to Romania           14/09/2011         RO         IMF Completes Second Review Under Stand-By Arrangement with Romania           07/11/2011         RO         Statement by the EC and IMF on the Review Mission to Romania           14/09/2011         RO         IMF Completes Second Review Under Stand-By Arrangement with Romania           07/11/2011         RO         Statement of the European Commission on the request by Hungary of possible financial assistance	01/04/2011	RO	Romania-Seventh Review Under the Stand-By Arrangement, Cancellation of Current Stand-By Arrangement, and Request for a New Stand-By Arrangement
11/04/2011         HU         Hungary—Post-Program Monitoring Discussions           09/05/2011         RO         European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania           09/06/2011         RO         IMF Letter of Intent and Technical Memorandum of Understanding           11/06/2011         RO         IMF Executive Board Concludes First Post-Program Monitoring Discussions and the Ex- Post Evaluation of Exceptional Access under the 2008 Stand-By Arrangement with Hungary           27/06/2011         RO         IMF Completes First Review Under Precautionary Stand-By Arrangement with Romania and approves 6481 Million Disbursement           29/06/2011         RO         The European Commission and Romania sign Memorandum of Understanding (MoU) on EUR 1.4 billion pre-cautionary balance-of-payments assistance           01/08/2011         RO         Statement by the EC and IMF on the Review Mission to Romania           14/09/2011         RO         IMF Completes Second Review Under Stand-By Arrangement with Romania           07/11/2011         RO         Statement of the European Commission on the request by Hungary of possible financial assistance           08/12/2011         HU         IMF Statement on Hungary           22/12/2011         RO         The Balance of Payments Programme for Romania's Economic Programme           05/02/2012         RO         Statement on Hungary           22/12/2011         RO	08/04/2011	HU	Commission's mission to Hungary welcomes commitment to structural reform and calls for further details on deficit cuts
09/05/2011         RO         European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania           09/06/2011         RO         IMF Letter of Intent and Technical Memorandum of Understanding           IMF Executive Board Concludes First Post-Program Monitoring Discussions and the Ex- 15/06/2011         HU         Post Evaluation of Exceptional Access under the 2008 Stand-By Arrangement with Hungary           27/06/2011         RO         IMF Completes First Review Under Precautionary Stand-By Arrangement with Romania and approves 6481 Million Disbursement           29/06/2011         RO         The European Commission and Romania sign Memorandum of Understanding (MoU) on EUR 1.4 billion pre-cautionary balance-of-payments assistance           01/08/2011         RO         Statement by the EC and IMF on the Review Mission to Romania           14/09/2011         RO         IMF Completes Second Review Under Stand-By Arrangement with Romania           07/11/2011         RO         Statement by the EC and IMF on the Review of Romania's Economic Program           21/11/2011         RO         Statement of the European Commission on the request by Hungary of possible financial assistance           08/12/2011         HU         IMF statement on Hungary           22/12/2011         RO         The Balance of Payments Programme for Romania's Economic Programme           05/02/2012         RO         Statement by the WB, EC and IMF on the Review of Romania's Economic	11/04/2011	HU	Hungary—Post-Program Monitoring Discussions
09/06/2011         RO         IMF Letter of Intent and Technical Memorandum of Understanding           15/06/2011         HU         IMF Executive Board Concludes First Post-Program Monitoring Discussions and the Ex- Post Evaluation of Exceptional Access under the 2008 Stand-By Arrangement with Hungary           27/06/2011         RO         IMF Completes First Review Under Precautionary Stand-By Arrangement with Romania and approves €481 Million Disbursement           29/06/2011         RO         The European Commission and Romania sign Memorandum of Understanding (MoU) on EUR 1.4 billion pre-cautionary balance-of-payments assistance           01/08/2011         RO         Statement by the EC and IMF on the Review Mission to Romania           14/09/2011         RO         IMF Completes Second Review Under Stand-By Arrangement with Romania           07/11/2011         RO         IMF Completes Second Review Under Stand-By Arrangement with Romania           07/11/2011         RO         IMF Completes Second Review Of Romania's Economic Program           21/11/2011         HU         Statement of the European Commission on the request by Hungary of possible financial assistance           08/12/2011         HU         IMF statement on Hungary           22/12/2011         RO         The Balance of Payments Programme for Romania. First Review - Autumn 2011           23/01/2012         RO         IMF Third Review Under the Stand-By Arrangement           05/02/2012	09/05/2011	RO	European Commission Staff Statement after a joint mission with the IMF and the World Bank to Romania
IMF       Executive Board Concludes First Post-Program Monitoring Discussions and the Ex- Post Evaluation of Exceptional Access under the 2008 Stand-By Arrangement with Hungary         27/06/2011       RO       IMF Completes First Review Under Precautionary Stand-By Arrangement with Romania and approves €481 Million Disbursement         29/06/2011       RO       The European Commission and Romania sign Memorandum of Understanding (MoU) on EUR 1.4 billion pre-cautionary balance-of-payments assistance         01/08/2011       RO       Statement by the EC and IMF on the Review Mission to Romania         14/09/2011       RO       IMF Completes Second Review Under Stand-By Arrangement with Romania         07/11/2011       RO       IMF Completes Second Review Under Stand-By Arrangement with Romania         07/11/2011       RO       IMF Completes Second Review Under Stand-By Arrangement with Romania         07/11/2011       RO       Statement by the EC and IMF on the Review of Romania's Economic Program         21/11/2011       HU       Statement of the European Commission on the request by Hungary of possible financial assistance         08/12/2011       HU       IMF statement on Hungary         22/12/2011       RO       IMF Third Review Under the Stand-By Arrangement         05/02/2012       RO       IMF Third Review Under Stand-By Arrangement for Romania's Economic Programme         28/02/2012       RO       IMF Completes Fourth Review Under Stand-By Arrang	09/06/2011	RO	IMF Letter of Intent and Technical Memorandum of Understanding
27/06/2011ROIMF Completes First Review Under Precautionary Stand-By Arrangement with Romania and approves €481 Million Disbursement29/06/2011ROThe European Commission and Romania sign Memorandum of Understanding (MoU) on EUR 1.4 billion pre-cautionary balance-of-payments assistance01/08/2011ROStatement by the EC and IMF on the Review Mission to Romania14/09/2011ROIMF Letter of Intent, and Technical Memorandum of Understanding29/09/2011ROIMF Completes Second Review Under Stand-By Arrangement with Romania07/11/2011ROStatement by the EC and IMF on the Review of Romania's Economic Program21/11/2011ROStatement of the European Commission on the request by Hungary of possible financial assistance08/12/2011HUIMF statement on Hungary22/12/2011ROThe Balance of Payments Programme for Romania. First Review - Autumn 201123/01/2012ROIMF Third Review Under the Stand-By Arrangement05/02/2012ROStatement by the WB, EC and IMF on the Review of Romania's Economic Programme28/02/2012ROLetter of Intent/ Memorandum of Economic and Financial Policies/ technical Memorandum of Understanding21/03/2012ROIMF Chinet/ Memorandum of Economic and Financial Policies/ technical Memorandum of Understanding21/03/2012ROIMF Completes Fourth Review Under Stand-By Arrangement for Romania25/04/2012HUIMF Completes Fourth Review Under Stand-By Arrangement for Romania11/03/2012ROIMF Completes Fourth Review Under Stand-By Arrangement for Romania21/03/2012<	15/06/2011	HU	IMF Executive Board Concludes First Post-Program Monitoring Discussions and the Ex- Post Evaluation of Exceptional Access under the 2008 Stand-By Arrangement with Hungary
29/06/2011ROThe European Commission and Romania sign Memorandum of Understanding (MoU) on EUR 1.4 billion pre-cautionary balance-of-payments assistance01/08/2011ROStatement by the EC and IMF on the Review Mission to Romania14/09/2011ROIMF Letter of Intent, and Technical Memorandum of Understanding29/09/2011ROIMF Completes Second Review Under Stand-By Arrangement with Romania07/11/2011ROStatement by the EC and IMF on the Review of Romania's Economic Program21/11/2011HUStatement of the European Commission on the request by Hungary of possible financial assistance08/12/2011HUIMF statement on Hungary22/12/2011ROThe Balance of Payments Programme for Romania. First Review - Autumn 201123/01/2012ROIMF Third Review Under the Stand-By Arrangement05/02/2012ROStatement by the WB, EC and IMF on the Review of Romania's Economic Programme28/02/2012ROIMF Third Review Under the Stand-By Arrangement21/03/2012ROIMF Completes Fourth Review Under Stand-By Arrangement for Romania21/03/2012ROIMF Executive Board Concludes Article IV Consultation and Second Post-Program Monitoring Discussions with Hungary + Commission decides to enter into negotiations on precautionary financial assistance with Hungary <td>27/06/2011</td> <td>RO</td> <td>IMF Completes First Review Under Precautionary Stand-By Arrangement with Romania and approves €481 Million Disbursement</td>	27/06/2011	RO	IMF Completes First Review Under Precautionary Stand-By Arrangement with Romania and approves €481 Million Disbursement
01/08/2011ROStatement by the EC and IMF on the Review Mission to Romania14/09/2011ROIMF Letter of Intent, and Technical Memorandum of Understanding29/09/2011ROIMF Completes Second Review Under Stand-By Arrangement with Romania07/11/2011ROStatement by the EC and IMF on the Review of Romania's Economic Program21/11/2011HUStatement of the European Commission on the request by Hungary of possible financial assistance08/12/2011HUIMF statement on Hungary22/12/2011ROThe Balance of Payments Programme for Romania. First Review - Autumn 201123/01/2012ROIMF Third Review Under the Stand-By Arrangement05/02/2012ROStatement by the WB, EC and IMF on the Review of Romania's Economic Programme28/02/2012ROLetter of Intent/ Memorandum of Economic and Financial Policies/ technical Memorandum of Understanding21/03/2012ROIMF Completes Fourth Review Under Stand-By Arrangement for Romania25/04/2012HUIMF Executive Board Concludes Article IV Consultation and Second Post-Program Monitoring Discussions with Hungary + Commission decides to enter into negotiations on precautionary financial assistance with Hungary	29/06/2011	RO	The European Commission and Romania sign Memorandum of Understanding (MoU) on EUR 1.4 billion pre-cautionary balance-of-payments assistance
14/09/2011ROIMF Letter of Intent, and Technical Memorandum of Understanding29/09/2011ROIMF Completes Second Review Under Stand-By Arrangement with Romania07/11/2011ROStatement by the EC and IMF on the Review of Romania's Economic Program21/11/2011HUStatement of the European Commission on the request by Hungary of possible financial assistance08/12/2011HUIMF statement on Hungary22/12/2011ROThe Balance of Payments Programme for Romania. First Review - Autumn 201123/01/2012ROIMF Third Review Under the Stand-By Arrangement05/02/2012ROStatement by the WB, EC and IMF on the Review of Romania's Economic Programme28/02/2012ROLetter of Intent/ Memorandum of Economic and Financial Policies/ technical Memorandum of Understanding21/03/2012ROIMF Completes Fourth Review Under Stand-By Arrangement for Romania25/04/2012HUMF Executive Board Concludes Article IV Consultation and Second Post-Program Monitoring Discussions with Hungary + Commission decides to enter into negotiations on precautionary financial assistance with Hungary	01/08/2011	RO	Statement by the EC and IMF on the Review Mission to Romania
29/09/2011ROIMF Completes Second Review Under Stand-By Arrangement with Romania07/11/2011ROStatement by the EC and IMF on the Review of Romania's Economic Program21/11/2011HUStatement of the European Commission on the request by Hungary of possible financial assistance08/12/2011HUIMF statement on Hungary22/12/2011ROThe Balance of Payments Programme for Romania. First Review - Autumn 201123/01/2012ROIMF Third Review Under the Stand-By Arrangement05/02/2012ROStatement by the WB, EC and IMF on the Review of Romania's Economic Programme28/02/2012ROStatement by the WB, EC and IMF on the Review of Romania's Economic Programme28/02/2012ROIMF Completes Fourth Memorandum of Economic and Financial Policies/ technical Memorandum of Understanding21/03/2012ROIMF Completes Fourth Review Under Stand-By Arrangement for RomaniaMonitoring Discussions with Hungary + Commission decides to enter into negotiations on precautionary financial assistance with Hungary	14/09/2011	RO	IMF Letter of Intent, and Technical Memorandum of Understanding
07/11/2011ROStatement by the EC and IMF on the Review of Romania's Economic Program21/11/2011HUStatement of the European Commission on the request by Hungary of possible financial assistance08/12/2011HUIMF statement on Hungary22/12/2011ROThe Balance of Payments Programme for Romania. First Review - Autumn 201123/01/2012ROIMF Third Review Under the Stand-By Arrangement05/02/2012ROStatement by the WB, EC and IMF on the Review of Romania's Economic Programme28/02/2012ROStatement by the WB, EC and IMF on the Review of Romania's Economic Programme28/02/2012ROIMF Third Nemorandum of Economic and Financial Policies/ technical Memorandum of Understanding21/03/2012ROIMF Completes Fourth Review Under Stand-By Arrangement for Romania25/04/2012HUMonitoring Discussions with Hungary + Commission decides to enter into negotiations on precautionary financial assistance with Hungary	29/09/2011	RO	IMF Completes Second Review Under Stand-By Arrangement with Romania
21/11/2011HUStatement of the European Commission on the request by Hungary of possible financial assistance08/12/2011HUIMF statement on Hungary22/12/2011ROThe Balance of Payments Programme for Romania. First Review - Autumn 201123/01/2012ROIMF Third Review Under the Stand-By Arrangement05/02/2012ROStatement by the WB, EC and IMF on the Review of Romania's Economic Programme28/02/2012ROLetter of Intent/ Memorandum of Economic and Financial Policies/ technical Memorandum of Understanding21/03/2012ROIMF Completes Fourth Review Under Stand-By Arrangement for Romania25/04/2012HUMonitoring Discussions with Hungary + Commission decides to enter into negotiations on precautionary financial assistance with Hungary	07/11/2011	RO	Statement by the EC and IMF on the Review of Romania's Economic Program
08/12/2011HUIMF statement on Hungary22/12/2011ROThe Balance of Payments Programme for Romania. First Review - Autumn 201123/01/2012ROIMF Third Review Under the Stand-By Arrangement05/02/2012ROStatement by the WB, EC and IMF on the Review of Romania's Economic Programme28/02/2012ROLetter of Intent/ Memorandum of Economic and Financial Policies/ technical Memorandum of Understanding21/03/2012ROIMF Completes Fourth Review Under Stand-By Arrangement for Romania25/04/2012HUMonitoring Discussions with Hungary + Commission decides to enter into negotiations on precautionary financial assistance with Hungary	21/11/2011	HU	Statement of the European Commission on the request by Hungary of possible financial assistance
22/12/2011ROThe Balance of Payments Programme for Romania. First Review - Autumn 201123/01/2012ROIMF Third Review Under the Stand-By Arrangement05/02/2012ROStatement by the WB, EC and IMF on the Review of Romania's Economic Programme28/02/2012ROLetter of Intent/ Memorandum of Economic and Financial Policies/ technical Memorandum of Understanding21/03/2012ROIMF Completes Fourth Review Under Stand-By Arrangement for Romania25/04/2012HUMonitoring Discussions with Hungary + Commission decides to enter into negotiations on precautionary financial assistance with Hungary	08/12/2011	HU	IMF statement on Hungary
23/01/2012       RO       IMF Third Review Under the Stand-By Arrangement         05/02/2012       RO       Statement by the WB, EC and IMF on the Review of Romania's Economic Programme         28/02/2012       RO       Letter of Intent/ Memorandum of Economic and Financial Policies/ technical Memorandum of Understanding         21/03/2012       RO       IMF Completes Fourth Review Under Stand-By Arrangement for Romania         25/04/2012       HU       IMF Executive Board Concludes Article IV Consultation and Second Post-Program Monitoring Discussions with Hungary + Commission decides to enter into negotiations on precautionary financial assistance with Hungary	22/12/2011	RO	The Balance of Payments Programme for Romania. First Review - Autumn 2011
05/02/2012ROStatement by the WB, EC and IMF on the Review of Romania's Economic Programme28/02/2012ROLetter of Intent/ Memorandum of Economic and Financial Policies/ technical Memorandum of Understanding21/03/2012ROIMF Completes Fourth Review Under Stand-By Arrangement for Romania25/04/2012HUIMF Executive Board Concludes Article IV Consultation and Second Post-Program Monitoring Discussions with Hungary + Commission decides to enter into negotiations on precautionary financial assistance with Hungary	23/01/2012	RO	IMF Third Review Under the Stand-By Arrangement
28/02/2012         RO         Letter of Intent/ Memorandum of Economic and Financial Policies/ technical Memorandum of Understanding           21/03/2012         RO         IMF Completes Fourth Review Under Stand-By Arrangement for Romania           25/04/2012         HU         IMF Executive Board Concludes Article IV Consultation and Second Post-Program Monitoring Discussions with Hungary + Commission decides to enter into negotiations on precautionary financial assistance with Hungary	05/02/2012	RO	Statement by the WB, EC and IMF on the Review of Romania's Economic Programme
21/03/2012         RO         IMF Completes Fourth Review Under Stand-By Arrangement for Romania           25/04/2012         HU         IMF Executive Board Concludes Article IV Consultation and Second Post-Program Monitoring Discussions with Hungary + Commission decides to enter into negotiations on precautionary financial assistance with Hungary	28/02/2012	RO	Letter of Intent/ Memorandum of Economic and Financial Policies/ technical Memorandum of Understanding
25/04/2012IMF Executive Board Concludes Article IV Consultation and Second Post-Program Monitoring Discussions with Hungary + Commission decides to enter into negotiations on precautionary financial assistance with Hungary	21/03/2012	RO	IMF Completes Fourth Review Under Stand-By Arrangement for Romania
	25/04/2012	HU	IMF Executive Board Concludes Article IV Consultation and Second Post-Program Monitoring Discussions with Hungary + Commission decides to enter into negotiations on precautionary financial assistance with Hungary

27/04/2012	RO	Joint press release by the IMF and EC Missions in Romania
09/05/2012	RO	Statement by the WB, EC, and the IMF on the Review of Romania's Economic Program
08/06/2012	RO	Letter of Intent/ Memorandum of Economic and Financial Policies/ Technical Memorandum of Understanding
22/06/2012	RO	IMF Completes Fifth Review Under Stand-By Arrangement for Romania
29/06/2012	RO	Second Supplemental Memorandum of Understanding between the EU and Romania
17/07/2012	HU	17 - 25 July: Mission to Hungary starts the negotiations on financial assistance
14/08/2012	RO	Statement by the IMF, EC and WB on the Review of Romania's Economic Program
12/09/2012	RO	Letter of Intent/ Memorandum of Economic and Financial Policies/ Technical Memorandum of Understanding
10/10/2012	RO	The Balance of Payments Programme for Romania. Second Review - Spring 2012
14/11/2012	RO	Statement of the IMF and EC Staff Visit
18/01/2013	PL	IMF renews Poland's Flexible Credit Line
28/01/2013	HU	16 - 28 January: Commission's mission to Hungary encourages continued progress in fiscal consolidation while paying more attention to raising growth potential
29/01/2013	RO	Statement of the IMF and EC Review Missions
20/03/2013	RO	IMF approves Three-Month Extension of SBA for Romania
29/03/2013	HU	IMF Executive Board Concludes 2013 Article IV Consultation and Third Post-Program Monitoring Discussions
10/06/2013	RO	IMF Letter of Intent, Memorandum of Economic and Financial Policies, and Technical Memorandum of Understanding
26/06/2013	RO	IMF Completes Seventh and Eighth Reviews Under the SBA for Romania
16/07/2013	RO	Romania: Seventh and Eighth Reviews Under the Stand-By Arrangement and Request for Waiver of Non-observance of Performance Criteria
19/07/2013	RO	Christine Lagarde to Visit Romania and Lithuania + Overall assessment of the two balance- of-payments assistance programmes for Romania, 2009-2013
31/07/2013	RO	Statement by the IMF and the EC on Joint Discussions on a New Economic Programme for Romania
12/08/2013	HU	Hungary Repays Early Its Outstanding Obligations to the IMF
12/09/2013	RO	Letter of Intent, Memorandum of Economic and Financial Policies, and Technical Memorandum of Understanding
27/09/2013	RO	IMF Executive Board approves New €1.98 Billion Precautionary Stand-By Arrangement for Romania
04/10/2013	RO	IMF Romania Request for a Stand-By Arrangement
29/10/2013	RO	Council Decision 2013/531/EU of 22 October 2013 providing precautionary Union medium-term financial assistance to Romania
05/11/2013	RO	Statement of the European Commission and International Monetary Fund Staff Visit
06/11/2013	RO	Memorandum of Understanding
20/11/2013	RO	Romania: Balance-of-Payments Assistance Programme 2013-2015
11/12/2013	HU	Commission staff conclude fourth Post-Programme Surveillance mission to Hungary
04/02/2014	RO	Romania: Statement at the Conclusion of the IMF and EC Staff Visit
05/03/2014	RO	Letter of Intent, Memorandum of Economic and Financial Policies, and Technical Memorandum of Understanding
26/03/2014	RO	IMF Executive Board Completes First and Second Reviews Under the Stand-By Arrangement and Ex-Post Evaluation of Exceptional Access for Romania
01/04/2014	RO	IMF Romania: First and Second Reviews Under the Stand-By Arrangement and Request for Waiver of Non-observance of a Performance Criterion
04/04/2014	RO	IMF Balance-of-Payments Assistance Programme
12/06/2014	RO	Statement at the conclusion of the IMF-European Commission staff visit to Romania
01/07/2014	HU	Commission staff conclude the fifth Post-Programme Surveillance mission to Hungary

Table 6B.	. Macroecono	mic releases	included in	ı the anal	ysis
-----------	--------------	--------------	-------------	------------	------

Czech Republic	Hungary	Poland	Romania
Av. Real Monthly Wage YoY	Av. Gross Wages YoY	Average Gross Wages MoM	CPI MoM
CPI MoM	Budget Balance YTD	Average Gross Wages YoY	CPI YoY
CPI YoY	CPI MoM	Budget Bal.: Performance YTD	GDP QoQ
Current Account (US\$)	CPI YoY	Budget Balance: Level YTD	GDP YoY
Current Account Balance	Current Account NSA	CPI Core MoM	
Current Account Monthly	GDP NSA YoY	CPI Core YoY	
GDP (constant prices) (YoY)	GDP SA QoQ	CPI MoM	
GDP QoQ	Ind. Production WDA YoY	CPI YoY	
GDP YoY	PPI MoM	Current Account Balance	
Industrial Output YoY	PPI YoY	Employment MoM	
Industrial Sales (YoY)	Retail Sales YoY	Employment YoY	
Manufacturing PMI	Trade Balance	Exports	
PPI Industrial MoM	Unemployment Rate	GDP Annual YoY	
PPI Industrial YoY		GDP QoQ	
Retail Sales YoY		GDP YoY	
Share of Unemployed 15-65		Imports	
Trade Balance		Manufacturing PMI	
Trade Balance NC		Money Supply M3 Level	
Unemployment Rate		Money Supply M3 MoM	
		Money Supply M3 YoY	
		NBP Inflation Expectations	
		Net Core Inflation (MoM)	
		Net Core Inflation (YoY)	
		PPI MoM	
		PPI YoY	
		Retail Sales MoM	
		Retail Sales YoY	
		Sold Industrial Output MoM	
		Sold Industrial Output YoY	
		Trade Balance	
		Unemployment Rate	

### Table 7B. Sovereign bonds denominated in foreign currency

Czech Republic	Hungary	Poland	Romania
XS0215153296	XS0212993678	XS0210314299	XS0371163600
	XS0240732114	XS0242491230	
	US445545AC05	US731011AP73	
	XS0219107918	JP561600A5B9	
	XS0249458984	JP561600A6B7	
		JP561600B6B6	

**Notes:** ISIN codes are reported. For Hungary and Poland we use the first principal component of the yields of the different bonds.

	LTRO FRTPFA							
	CZ	HU	PL	RO	CZ	HU	PL	RO
Exchange rate	-0.21**	0.27*	-0.13	0.23**	0.01	-0.37*	-0.11	-0.36***
	(0.10)	(0.15)	(0.14)	(0.09)	(0.14)	(0.21)	(0.19)	(0.12)
Stock market index	0.30	-0.96*	0.34	0.49	0.11	2.58***	0.02	-0.33
	(0.35)	(0.50)	(0.34)	(0.37)	(0.47)	(0.67)	(0.45)	(0.49)
Interbank rate (3M)	0.62***	0.27	0.17*	-0.61	-0.88***	-0.16	0.08	0.24
	(0.23)	(0.21)	(0.10)	(0.49)	(0.31)	(0.28)	(0.13)	(0.65)
Benchmark bond yield (5Y; LC)	2.34	-0.80	0.92	6.11	-7.60**	3.38	0.00	-10.10
	(2.27)	(3.00)	(1.42)	(10.57)	(3.06)	(3.97)	(1.88)	(14.01)
Benchmark bond yield (10Y; LC)	-2.01*	-4.33	-0.53	3.05	0.29	5.01	-0.84	-7.97
	(1.21)	(2.81)	(1.37)	(4.39)	(1.63)	(3.72)	(1.82)	(5.82)
Bond yield (FX)	-2.08**	-4.05**	-1.01	-1.12	0.87	-2.34	-1.78	0.33
	(0.83)	(1.88)	(1.69)	(2.31)	(1.13)	(2.49)	(2.24)	(2.74)
CDS 5-year (USD)	0.90	0.96	-0.29	0.15	-0.36	-0.18	0.27	-0.54
	(1.14)	(0.81)	(0.96)	(0.68)	(1.53)	(1.08)	(1.27)	(0.90)
CDS 10-year (USD)	0.54	1.41*	-0.68	0.34	0.28	-1.20	-0.08	-1.82**
	(1.04)	(0.77)	(0.85)	(0.67)	(1.40)	(1.02)	(1.13)	(0.88)

Table 8B. Event-study analysis on ECB's monetary policy decisions (LTRO and FRTPFA, 1-day window)

**Note:** In addition to variables noted in the equation in Section 5.1, the regressions include two interaction terms: one for domestic and ECB key policy rate changes and the one for domestic and ECB non-standard monetary policy measures. In addition, we also include a dummy variable for the exchange rate regression for the Czech Republic in order to account for a regime change since 9 November 2013 (i.e. since when the CNB intervened and declared that the 27 CZK per EUR level is the preferred one). The  $\delta$ -coefficients from equation (1) for the respective ECB non-standard monetary policy measure are reported in this table jointly with standard errors in parentheses; \*\*\* p<0.01, \*\* p<0.05, \* p<0.1

Source: Authors' calculations.

			FOR		COLL			
	CZ	HU	PL	RO	CZ	HU	PL	RO
Exchange rate	-0.04	-0.05	0.19*	0.10	0.16	0.49**	0.18	-0.22
	(0.08)	(0.12)	(0.11)	(0.07)	(0.17)	(0.25)	(0.23)	(0.15)
Stock market index	-0.45*	0.01	-0.47*	-0.18	1.89***	0.71	0.61	1.23**
	(0.20)	(0.58)	(0.23)	(0.28)	(0.50)	(0.81)	(0.55)	(0.00)
Interbank rate (3M)	-0.16	0.25	-0.19**	-0.62*	-0.93**	-0.08	-0.02	0.49
	(0.17)	(0.16)	(0.07)	(0.37)	(0.37)	(0.34)	(0.16)	(0.79)
Benchmark bond yield (5Y; LC)	-0.77	-4.27*	0.32	-4.97	1.86	8.65*	5.28**	8.19
•	(1.70)	(2.25)	(1.07)	(7.93)	(3.66)	(4.83)	(2.30)	(17.05)
Benchmark bond yield (10Y; LC)	-1.92**	-0.73	-3.53***	-12.66***	2.02	13.26***	5.13**	4.48
	(0.90)	(2.11)	(1.03)	(3.30)	(1.95)	(4.52)	(2.23)	(7.09)
Bond yield (FX)	-1.65***	1.35	1.57	2.44	1.27	4.07	3.36	11.09***
	(0.62)	(1.41)	(1.27)	(1.50)	(1.35)	(3.03)	(2.74)	(3.07)
CDS 5-year (USD)	0.51	0.16	0.82	0.14	4.79***	5.98***	5.06***	2.78**
	(0.85)	(0.61)	(0.72)	(0.51)	(1.83)	(1.31)	(1.55)	(1.09)
CDS 10-year (USD)	-0.15	-0.02	-0.22	-0.00	4.89***	5.76***	1.72	3.03***
• • •	(0.77)	(0.58)	(0.64)	(0.50)	(1.67)	(1.23)	(1.38)	(1.08)

#### Table 9B. Event-study analysis on ECB's monetary policy decisions (FOR and COLL, 1-day window)

Note: In addition to variables noted in the equation in Section 5.1, the regressions include two interaction terms: one for domestic and ECB key policy rate changes and the one for domestic and ECB non-standard monetary policy measures. In addition, we also include a dummy variable for the exchange rate regression for the Czech Republic in order to account for a regime change since 9 November 2013 (i.e. since when the CNB intervened and declared that the 27 CZK per EUR level is the preferred one). The  $\delta$ -coefficients from equation (1) for the respective ECB non-standard monetary policy measure are reported in this table jointly with standard errors in parentheses; \*\*\* p<0.01, \*\* p<0.05, \* p<0.1

Source: Authors' calculations.

		PP		FWG				
	CZ	HU	PL	RO	CZ	HU	PL	RO
Exchange rate	0.01	0.48	0.03	-0.09	-0.02	0.26	-0.76	-0.32
	(0.20)	(0.29)	(0.27)	(0.17)	(0.40)	(0.60)	(0.55)	(0.35)
Stock market index	1.14*	-0.58	0.10	0.57	-1.53	0.94	0.13	-0.48
	(0.65)	(0.94)	(0.63)	(0.69)	(0.96)	(1.40)	(0.93)	(1.02)
Interbank rate (3M)	-0.41	-0.00	0.01	-0.69	0.08	-0.18	-0.74*	-0.11
	(0.44)	(0.40)	(0.18)	(0.91)	(0.90)	(0.83)	(0.38)	(1.89)
Benchmark bond yield (2/3Y; LC)	-16.85***	8.47	0.68	-5.06	-3.55	-0.89	-8.26	-1.33
	(4.30)	(5.61)	(2.65)	(19.78)	(8.79)	(11.59)	(5.49)	(40.92)
Benchmark bond yield (10Y; LC)	2.45	7.10	-1.45	4.06	-9.97**	-1.49	-13.44**	0.53
	(2.29)	(5.25)	(2.57)	(8.22)	(4.69)	(10.85)	(5.31)	(17.01)
Bond yield (FX)	5.33***	6.11*	3.68	2.76	-1.00	-0.48	0.03	0.72
	(1.58)	(3.52)	(3.16)	(3.54)	(3.23)	(7.27)	(6.53)	(7.29)
CDS 5-year (USD)	0.60	-2.92*	-1.02	-1.60	0.97	3.39	2.52	-0.39
	(2.15)	(1.52)	(1.79)	(1.27)	(4.40)	(3.14)	(3.71)	(2.63)
CDS 10-year (USD)	0.63	-2.98**	-1.26	-1.30	0.40	3.26	2.29	-0.48
	(1.96)	(1.43)	(1.59)	(1.25)	(4.01)	(2.96)	(3.30)	(2.58)

Table 10B. Event-study analysis on ECB's monetary policy decisions (CBPP and FWG, 1-day window)

Note: In addition to variables noted in the equation in Section 5.1, the regressions include two interaction terms: one for domestic and ECB key policy rate changes and the one for domestic and ECB non-standard monetary policy measures. In addition, we also include a dummy variable for the exchange rate regression for the Czech Republic in order to account for a regime change since 9 November 2013 (i.e. since when the CNB intervened and declared that the 27 CZK per EUR level is the preferred one). The  $\delta$ -coefficients from equation (1) for the respective ECB non-standard monetary policy measure are reported in this table jointly with standard errors in parentheses; \*\*\* p<0.01, \*\* p<0.05, \* p<0.1

Source: Authors' calculations.

		ABS	SPP		TLTRO				
	CZ	HU	PL	RO	CZ	HU	PL	RO	
Exchange rate	-0.06 (0.29)	-0.84* (0.43)	-0.25	0.06	0.09 (0.22)	0.14 (0.33)	-0.18 (0.31)	-0.01	
Stock market index	-1.53	0.94	0.13	-0.48	-0.63	-0.59	-0.08	-0.35	
	(0.96)	(1.40)	(0.93)	(1.02)	(0.74)	(1.09)	(0.72)	(0.80)	
Interbank rate (3M)	0.76	-0.15	-0.43	0.76	0.20	0.05	-0.04	-0.77	
	(0.65)	(0.59)	(0.27)	(1.35)	(0.50)	(0.46)	(0.21)	(1.05)	
Benchmark bond yield (2/3Y;	12.64**	-11.93	1.62	-5.66	0.67	6.69	-2.95	-6.48	
	(6.35)	(8.29)	(3.93)	(29.26)	(4.90)	(6.46)	(3.05)	(22.76)	
Benchmark bond yield (10Y;	-3.12	-19.31**	1.85	-3.70	3.63	11.65*	-2.87	-1.08	
	(3.39)	(7.76)	(3.80)	(12.16)	(2.61)	(6.05)	(2.96)	(9.46)	
Bond yield (FX)	-4.98**	-2.69	-3.03	-8.18	-0.54	7.58*	1.35	-0.04	
	(2.33)	(5.20)	(4.67)	(5.23)	(1.80)	(4.05)	(3.63)	(4.06)	
CDS 5-year (USD)	-2.00	-1.92	-1.70	-1.20	-0.83	-1.14	-0.75	0.88	
	(3.18)	(2.24)	(2.65)	(1.88)	(2.45)	(1.75)	(2.06)	(1.46)	
CDS 10-year (USD)	-1.97	-1.26	-0.19	-0.82	-0.73	-0.95	0.04	0.93	
	(2.89)	(2.12)	(2.36)	(1.85)	(2.23)	(1.65)	(1.83)	(1.44)	

## Table 11B. Event-study analysis on ECB's monetary policy decisions (ABSPP and TLTRO, 1-day window)

Note: In addition to variables noted in the equation in Section 5.1, the regressions include two interaction terms: one for domestic and ECB key policy rate changes and the one for domestic and ECB non-standard monetary policy measures. In addition, we also include a dummy variable for the exchange rate regression for the Czech Republic in order to account for a regime change since 9 November 2013 (i.e. since when the CNB intervened and declared that the 27 CZK per EUR level is the preferred one). The  $\delta$ -coefficients from equation (1) for the respective ECB non-standard monetary policy measure are reported in this table jointly with standard errors in parentheses; \*\*\* p<0.01, \*\* p<0.05, \* p<0.1

Source: Authors' calculations.





Source: see Appendix A for Figures 1B-9B. Authors' calculations based on equation (1) for Figure 10B.



Notes: Eight major events are depicted. 1. First 6-month LTROs (28/03/2008); 2. Enhanced Credit Support (07/05/2009); 3. SMP (10/05/2010); 4. Active implementation of the SMP (08/08/2011); 5. 3-year LTROs (08/12/2011); 6. Draghi's London speech (26/07/2012); 7. TLTROs (05/06/2014); 8. Draghi's speech at the EP (17/11/2014). Source: Authors' calculations based on equation (1).



Figure 12B. Rolling regressions (Hungary)

Notes: Eight major events are depicted. 1. First 6-month LTROs (28/03/2008); 2. Enhanced Credit Support (07/05/2009); 3. SMP (10/05/2010); 4. Active implementation of the SMP (08/08/2011); 5. 3-year LTROS (08/12/2011); 6. Draghi's London speech (26/07/2012); 7. TLTROs (05/06/2014); 8. Draghi's speech at the EP (17/11/2014). Source: Authors' calculations based on equation (1).



Figure 13B. Rolling regressions (Poland)

Notes: Eight major events are depicted. 1. First 6-month LTROs (28/03/2008); 2. Enhanced Credit Support (07/05/2009); 3. SMP (10/05/2010); 4. Active implementation of the SMP (08/08/2011); 5. 3-year LTROS (08/12/2011); 6. Draghi's London speech (26/07/2012); 7. TLTROs (05/06/2014); 8. Draghi's speech at the EP (17/11/2014). Source: Authors' calculations based on equation (1).



Notes: Eight major events are depicted. 1. First 6-month LTROs (28/03/2008); 2. Enhanced Credit Support (07/05/2009); 3. SMP (10/05/2010); 4. Active implementation of the SMP (08/08/2011); 5. 3-year LTROs (08/12/2011); 6. Draghi's London speech (26/07/2012); 7. TLTROs (05/06/2014); 8. Draghi's speech at the EP (17/11/2014). Source: Authors' calculations based on equation (1).

## **APPENDIX C: Weighted dummies**

Following up on Rogers et al. (2014), we use the spread between the Italian and German 10year sovereign bond yields in order to measure the size of monetary policy surprise stemming from the ECB announcements and we use it as an alternative explanatory variable  $NSMP_t^{ECB}$ in our event-study analysis. We transform this measure so that its positive value indicates a positive surprise stemming from the ECB announcements, namely a compression of the 10year sovereign bonds yield. The results are broadly consistent with those using the binary dummy variable albeit with some interesting divergences.

Most notably, the ECB announcements are associated with a statistically significant appreciation in local currencies vis-à-vis the euro, and a decline in CDS across all countries in our sample (see Table 1C). These spillovers are somewhat stronger compared with the ones obtained from our baseline specification using the binary dummy variable. In line with our earlier results, we find evidence of notable spillovers from ECB announcements on sovereign bond yields. Analysing the selected ECB announcements of non-standard monetary policy measures separately (see Table 2C), our baseline results – strong spillovers from the SMP and more muted ones from the OMT and the PSPP – are confirmed.

	ECB's non-standard policies						
	CZ	CZ	HU	PL			
Exchange rate	-0.01***	-1.10***	-0.73*	-1.04***			
-	(0.00)	(0.29)	(0.43)	(0.40)			
Stock market index	-0.01	-2.58***	-4 16***	-2.02**			
	(0.01)	(0.97)	(1.41)	(0.94)			
Interbank rate (3M)	0.02***	1.04	0.41	0.23			
interbank fate (SWI)	(0.01)	(0.65)	(0.60)	(0.23)			
Banahmark hand yield (2/2V, IC)	0.02	0.77	20 15***	10.05***			
Benchmark bond yield (2/3 Y; LC)	-0.03	-3.77	-28.15***	-12.35***			
	(0.07)	(6.38)	(8.40)	(3.97)			
Benchmark bond yield (10Y; LC)	-0.03	1.30	-31.50***	-3.91			
-	(0.04)	(3.40)	(7.86)	(3.84)			
Bond vield (FX)	0.01	0.19	-7.86	-8.24*			
	(0.02)	(2.35)	(5.27)	(4.73)			
CDS 5-year (USD)	0.12***	13 53***	10.05***	12 00***			
655 5 year (655)	(0.03)	(3.19)	(2.27)	(2.68)			
CDS 10-year (USD)	-0.13***	-13.06***	-11.17***	-11.55***			
	(0.03)	(2.91)	(2.14)	(2.39)			

Table 1C. Event-study analysis on ECB's monetary policy decisions (1-day window)

**Note:** In addition to variables noted in the equation in Section 5.1, the regressions include two interaction terms: one for domestic and ECB key policy rate changes and the one for domestic and ECB non-standard monetary policy measures. In addition, we also include a dummy variable for the exchange rate regression for the Czech Republic in order to account for a regime change since 9 November 2013 (i.e. since when the CNB intervened and declared that the 27 CZK per EUR level is the preferred one). The  $\delta$ -coefficients from equation (1) for the respective ECB non-standard monetary policy measure are reported in this table jointly with standard errors in parentheses; \*\*\* p<0.01, \*\* p<0.05, \* p<0.1 **Source:** Authors' calculations.

	SMP			OMT				PSPP				
	CZ	HU	PL	RO	CZ	HU	PL	RO	CZ	HU	PL	RO
Exchange rate	-0.02***	-0.01	-0.01**	0.00	-0.01	-0.00	-0.00	-0.01	0.03	-0.02	-0.03	-0.02
	(0.00)	(0.01)	(0.01)	(0.00)	(0.01)	(0.01)	(0.01)	(0.01)	(0.04)	(0.06)	(0.06)	(0.04)
Stock market index	-0.04***	-0.06***	-0.03**	-0.07***	-0.02	-0.00	-0.01	-0.01	0.03	0.10	0.23*	0.01
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)	(0.03)	(0.02)	(0.02)	(0.14)	(0.21)	(0.14)	(0.15)
Interbank rate (3M)	-0.02	-0.01	-0.00	-0.09***	-0.03**	0.00	0.00	-0.01	-0.01	0.00	0.04	0.05
	(0.01)	(0.01)	(0.00)	(0.02)	(0.01)	(0.01)	(0.01)	(0.03)	(0.10)	(0.09)	(0.04)	(0.20)
Benchmark bond yield (2/3Y; LC)	-0.04	-0.39***	-0.18***	-0.43	0.06	-0.03	-0.09	-0.15	0.06	0.08	0.01	0.26
	(0.10)	(0.13)	(0.06)	(0.48)	(0.13)	(0.18)	(0.09)	(0.62)	(0.95)	(1.25)	(0.59)	(4.42)
Benchmark bond yield (10Y; LC)	0.02	-0.44***	-0.04	-0.64***	-0.02	-0.06	-0.15*	-0.29	0.37	-0.15	-0.28	0.29
	(0.05)	(0.13)	(0.06)	(0.20)	(0.07)	(0.17)	(0.08)	(0.26)	(0.50)	(1.17)	(0.57)	(1.84)
Bond yield (FX)	0.00	-0.09	-0.10	-0.05	-0.02	0.05	-0.01	-0.09	0.19	2.03***	0.12	0.03
	(0.04)	(0.08)	(0.08)	(0.09)	(0.05)	(0.11)	(0.10)	(0.11)	(0.35)	(0.79)	(0.71)	(0.79)
CDS 5-year (USD)	-0.17***	-0.15***	-0.16***	-0.14***	-0.06	-0.06	-0.09	-0.05	-0.09	0.17	-0.13	0.31
	(0.05)	(0.04)	(0.04)	(0.03)	(0.07)	(0.05)	(0.06)	(0.04)	(0.47)	(0.34)	(0.40)	(0.28)
CDS 10-year (USD)	-0.16***	-0.15***	-0.15***	-0.14***	-0.07	-0.06	-0.07	-0.05	-0.06	0.16	-0.13	0.24
	(0.05)	(0.03)	(0.04)	(0.03)	(0.06)	(0.05)	(0.05)	(0.04)	(0.43)	(0.32)	(0.36)	(0.28)
	SM	IP announceme	ent (10 May 20	)10)	OMT	OMT (Draghi's speech, 26 July 2012)			PSPP announcement (22 January 2015)			
	CZ	HU	PL	RO	CZ	HU	PL	RO	CZ	HU	PL	RO
Exchange rate	-0.04***	-0.02	-0.04***	0.00	-0.01	-0.01	-0.01	-0.00	0.03	-0.07	-0.07	-0.04
	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)	(0.07)	(0.10)	(0.09)	(0.06)
Stock market index	-0.05*	-0.13***	-0.05*	-0.10***	0.00	0.01	-0.01	0.04	0.18	0.40	0.37*	0.05
	(0.03)	(0.04)	(0.03)	(0.03)	(0.03)	(0.04)	(0.03)	(0.03)	(0.23)	(0.33)	(0.22)	(0.24)
Interbank rate (3M)	-0.02	-0.02	-0.01	-0.34***	0.00	-0.00	0.00	-0.01	-0.02	0.00	0.01	0.22
	(0.02)	(0.02)	(0.01)	(0.04)	(0.02)	(0.02)	(0.01)	(0.04)	(0.15)	(0.14)	(0.06)	(0.32)
Benchmark bond yield (2/3Y; LC)	-0.22	-0.96***	-0.37***	-1.93**	-0.06	0.05	-0.15	-0.00	0.11	0.60	-0.68	0.57
	(0.18)	(0.24)	(0.11)	(0.86)	(0.20)	(0.27)	(0.13)	(0.94)	(1.48)	(1.96)	(0.93)	(6.91)
Benchmark bond yield (10Y; LC)	0.06	-1.08***	-0.27**	-2.08***	-0.01	-0.04	-0.28**	0.03	0.83	0.40	-0.74	0.49
	(0.10)	(0.23)	(0.11)	(0.36)	(0.11)	(0.25)	(0.13)	(0.39)	(0.79)	(1.83)	(0.90)	(2.86)
Bond yield (FX)	0.00	-0.44***	-0.38***	-0.18	-0.17**	-0.02	-0.01	-0.09	0.44	4.83***	0.23	0.61
	(0.07)	(0.15)	(0.14)	(0.16)	(0.07)	(0.17)	(0.16)	(0.17)	(0.54)	(1.23)	(1.10)	(1.23)
CDS 5-year (USD)	-0.60***	-0.43***	-0.49***	-0.39***	-0.03	-0.05	-0.01	-0.03	-0.42	0.27	-0.24	0.41
	(0.09)	(0.07)	(0.08)	(0.05)	(0.10)	(0.07)	(0.09)	(0.06)	(0.74)	(0.53)	(0.62)	(0.44)
CDS 10-year (USD)	-0.58***	-0.44***	-0.48***	-0.40***	-0.04	-0.05	-0.03	-0.03	-0.26	0.27	-0.26	0.33
	(0.08)	(0.06)	(0.07)	(0.05)	(0.09)	(0.07)	(0.08)	(0.06)	(0.67)	(0.50)	(0.55)	(0.43)

Table 2C. Event-study analysis on ECB's monetary policy decisions (1
--

Note: In addition to variables noted in the equation in Section 5.1, the regressions include two interaction terms: one for domestic and ECB key policy rate changes and the one for domestic and ECB non-standard monetary policy measures. In addition, we also include a dummy variable for the exchange rate regression for the Czech Republic in order to account for a regime change since 9 November 2013 (i.e. since when the CNB intervened and declared that the 27 CZK per EUR level is the preferred one). The  $\delta$ -coefficients from equation (1) for the respective ECB non-standard monetary policy measure are reported in this table jointly with standard errors in parentheses; \*\*\* p<0.01, \*\* p<0.05, \* p<0.1

Source: Authors' calculations.

#### Acknowledgements

The views expressed in this paper are those of the authors and do not necessarily reflect those of the ECB, the Eurosystem or the European Commission. We thank Carlo Altavilla, Søren Autrup, Pavel Gertler, Michele Lenza, Martin Schmitz, and an anonymous referee for valuable suggestions. We also thank participants at ECB seminars, the 1st International Research Conference on "Macroeconomic Policies and Financial Stability Issues in Emerging Markets" at the Central Bank of the Republic of Azerbaijan, the 17th Annual DNB Research Conference on "Forward Guidance and Communication about Unconventional Monetary Policy" at De Nederlandsche Bank and the 14th Annual European Economics and Finance Society Annual Conference at the Centre for European Policy Studies.

#### Matteo Falagiarda (corresponding author)

European Central Bank, Frankfurt am Main, Germany; e-mail: Matteo.Falagiarda@ecb.int

#### Peter McQuade

European Commission, Brussels, Belgium; e-mail: Peter.McQuade@ec.europa.eu

#### Marcel Tirpák

European Central Bank, Frankfurt am Main, Germany; e-mail: Marcel.Tirpak@ecb.int

#### © European Central Bank, 2015

Postal address60640 Frankfurt am Main, GermanyTelephone+49 69 1344 0Websitewww.ecb.europa.eu

All rights reserved. Any reproduction, publication and reprint in the form of a different publication, whether printed or produced electronically, in whole or in part, is permitted only with the explicit written authorisation of the ECB or the authors.

This paper can be downloaded without charge from www.ecb.europa.eu, from the Social Science Research Network electronic library at http://ssrn.com or from RePEc: Research Papers in Economics at https://ideas.repec.org/s/ecb/ecbwps.html.

Information on all of the papers published in the ECB Working Paper Series can be found on the ECB's website, http://www.ecb.europa.eu/pub/scientific/wps/date/html/index.en.html.

ISSN	1725-2806 (online)
ISBN	978-92-899-1682-0
DOI	10.2866/676909
EU catalogue No	QB-AR-15-109-EN-N